The effects of integrated reporting quality: a meta-analytic review

Giulia Zennaro, Giulio Corazza and Filippo Zanin Department of Economics and Statistics, University of Udine, Udine, Italy

Abstract

Purpose – The effects of integrated reporting quality (IRQ) have been debated in increasing empirical studies. Several IRQ measures, different theoretical approaches and multiple contexts have been adopted and investigated, leading to mixed results. By using the meta-analytic technique, this study aims to contribute to the accounting literature, reconciling the conflicting results on the effects of IRQ and providing objective conclusions to complement narrative literature reviews.

Design/methodology/approach – A sample of 45 empirical papers from 2013 to 2022, with 653 effect sizes, was used to assess the effects associated with IRQ. The papers were clustered into five groups (market reaction, financial performance, cost of capital, financial analysts' properties and managerial decisions) based on the different consequences of IRQ investigated in the primary studies. A random-effects meta-regression model was used to explore all sources of heterogeneity together.

Findings – The meta-regression results confirm that IRQ positively influences firms' market valuation and financial performance and hampers opportunistic managerial behaviour by improving corporate transparency, mitigating information asymmetry and encouraging accountability. Moreover, differences in the study characteristics affect the strength of the relationship object of interest.

Originality/value – Through meta-analysis, this study provides a broader overview of the effects of IRQ by enhancing the generalisability of the findings. The results also pave the way for additional evidence on the outcome variables affected by the quality of integrated disclosure.

Keywords Integrated reporting, Integrated reporting quality effects, Meta-analysis

Paper type Literature review

1. Introduction

Recent events, such as the global financial crisis, unfair corporate behaviour, scandals and raised awareness of social and environmental issues, have highlighted the pitfalls of traditional corporate reporting. This led to a more extensive integration of financial and non-financial information through integrated reporting (IR) (Vitolla *et al.*, 2019), characterised by a particular focus on how value is created, distributed and communicated. Despite the success of this innovative corporate reporting tool and the growing number of studies in this field, the results on the effects associated with integrated disclosure quality are still inconclusive (Nwachukwu, 2022; Hossain *et al.*, 2023).

© Giulia Zennaro, Giulio Corazza and Filippo Zanin. Published by Emerald Publishing Limited. This article is published under the Creative Commons Attribution (CC BY 4.0) licence. Anyone may reproduce, distribute, translate and create derivative works of this article (for both commercial and non-commercial purposes), subject to full attribution to the original publication and authors. The full terms of this licence may be seen at http://creativecommons.org/licences/by/4.0/legalcode

Giulia Zennaro's scholarship is funded by: Municipal Doctoral scholarship with Ministerial Decree 725/2021 and Decrees of the Agency for Territorial Cohesion no. 185/2021 and no. 222/2021. Giulio Corazza's contract is funded by the European Union – FSE REACT-EU, PON Research and Innovation 2014-2020.

Meditari Accountancy Research

197

Received 30 September 2023 Revised 26 March 2024 17 June 2024 22 July 2024 Accepted 23 July 2024



Meditari Accountancy Research Vol. 32 No. 7, 2024 pp. 197-235 Emerald Publishing Limited 2049-372X DOI 10.1108/MEDAR-09-2023-2175

MEDAR 32,7

This study aims to review empirical papers on this topic using the meta-analytic technique, which allows combining and synthesising conflicting research findings to obtain an objective view of the consequences of integrated reporting quality (IRQ), considering the sources of heterogeneity among the papers, such as differences in study designs and research settings. Meta-analysis enables an increase in statistical power and validity that may be absent from individual studies (Ahamed *et al.*, 2023), as it reduces the weight of papers with small sample sizes in determining the overall relationship, considers the discrepancies among them and promotes results generalisation (Walker *et al.*, 2008; Hunter and Schmidt, 2014). According to Lipsey and Wilson (2001), in our meta-analysis, Fisher's Z transformation is applied to all the effect size estimates, and the meta-regression model is used to explore the sources of heterogeneity across studies (measurement choices, control variables and study characteristics) and draw conclusions on the effects of IRO.

The literature exploring the effects of integrated disclosure quality is developing worldwide (Crous *et al.*, 2022; Hossain *et al.*, 2023). Many empirical studies, adopting several theoretical frameworks, investigated the market reactions to the disclosure of high-quality information in an integrated manner (Lee and Yeo, 2016; Barth *et al.*, 2017; Sriani and Agustia, 2020; Abogazia *et al.*, 2022; Soriya and Rastogi, 2022), while some others focused on the impact of the enhanced quality of IR on financial performance (Pavlopoulos *et al.*, 2019; Adegboyegun *et al.*, 2020; Soriya and Rastogi, 2022), the cost of capital (Vitolla *et al.*, 2020; Chouaibi *et al.*, 2022; Raimo *et al.*, 2022), the firm's information environment (Zhou *et al.*, 2017; Zúñiga *et al.*, 2020; Leukhardt *et al.*, 2022) and managerial decisions (Donkor *et al.*, 2022; Wu and Zhou, 2022).

From the analysis of 45 articles with 653 effect sizes (ES), this review aims to contribute to the accounting literature and the non-financial reporting stream of research, reconciling the conflicting results on the effects of IRQ and providing objective conclusions that may be unclear from narrative literature reviews. To the best of the authors' knowledge, no one else has used metaanalysis to investigate this relationship. This represents an important gap; many researchers have shown interest towards IR but have obtained mixed evidence (Hossain *et al.*, 2023).

Despite its widespread diffusion in other research streams, such as medicine and psychology, the meta-analysis technique is rarely applied in the accounting literature, even if its relevance has increased recently (Pomeroy and Thornton, 2008; Velte, 2019). Prior metaanalyses in this field have been conducted mainly on corporate governance issues (García-Meca and Sánchez-Ballesta, 2010; Siddiqui, 2015; Majumder *et al.*, 2017; Wang and Shailer, 2018), external auditing (Hay *et al.*, 2006; Habib, 2012; Ng *et al.*, 2018), management accounting (Derfuss, 2009; Liu *et al.*, 2014) and financial accounting (Khlif and Souissi, 2010; Ahmed *et al.*, 2013; Mokhtar, 2017); however, few meta-analyses have addressed the topic of non-financial reporting (Velte, 2019).

The contributions of this work are manifold: firstly, it applies a rigorous meta-analytical methodology to the specific topic of IR that, up to this moment, to the best of our knowledge, has only been adopted by Dragomir and Dumitru (2023), who investigated the relationship between corporate governance and IRQ. Secondly, even after broadening the spectrum of analysis to include all literature reviews that have analysed this topic using different approaches, few studies have addressed the issue of IRQ effects (Vitolla *et al.*, 2019; Crous *et al.*, 2022; Nwachukwu, 2022; Hossain *et al.*, 2023). Therefore, this work complements the narrative reviews on this topic. Finally, this paper contributes to the debate on IRQ by reconciling conflicting results, providing generalisable findings and opening new avenues for future research.

The remainder of this paper is organised as follows. Section 2 reviews the relevant research evidence and develops hypotheses related to the effects of IRQ. Section 3 describes the meta-analytic technique and the characteristics of the selected studies. Section 4 presents and analyses the results of the meta-analysis. In Section 5, we discuss the results obtained in the previous section. Finally, Section 6 concludes the paper.

2. Theoretical background and hypotheses development

This section reviews the literature on IRQ, discusses the different IRQ assessment methods and analyses IRQ effects to reconcile the conflicting results scholars in this stream of research obtained.

2.1 Integrated reporting and integrated reporting quality measurement

Since its introduction, IR has attracted the interest of academics and practitioners as an innovative tool (Stubbs and Higgins, 2014; Vitolla et al., 2019; Songini et al., 2022) that promotes a comprehensive and cohesive approach to corporate reporting (Hoang *et al.*, 2020; IIRC, 2021). An integrated report is "a concise communication about how an organisation's strategy, governance, performance and prospects, in the context of its external environment, lead to the creation, preservation or erosion of value over the short, medium and long term" (IIRC, 2021, p. 10). It provides a holistic representation of the business and its ability to create value in the short, medium and long term as a combination of six different forms of capital, namely financial, manufactured, intellectual, human, social and relationship and natural capital (IIRC, 2021). Beyond the external reporting format, the IR process drives internal change in decision-making through integrated thinking, an innovative strategic and cultural approach that enables managers to break down organisational "silos" (Al-Htaybat and von Alberti-Alhtaybat, 2018; Veltri and Silvestri, 2020; Busco et al., 2021; De Villiers and Dimes, 2023). Given its emphasis on the connectivity of information and focus on the factors affecting the value creation process (IIRC, 2021), recent literature argued that integrated reports play a key role in bridging the information gap for capital providers and other stakeholders and supported their ability to meet various non-financial and sustainability reporting requirements, in the context of emerging multiple reporting frameworks and solutions (García-Sánchez and Noguera-Gámez, 2017; Al-Htaybat and von Alberti-Alhtavbat, 2018: Maroun, 2022: De Villiers and Dimes, 2023).

However, IR adoption has severe limitations in capturing actual IR practices (Hossain *et al.*, 2023); it is crucial also to investigate IRQ (Velte, 2022), which is a critical aspect of corporate disclosure, although the concept of quality is not easy to define. In 2013, the International Integrated Reporting Council (IIRC) (in 2021, it merged with the Sustainability Accounting Standards Board to form the Value Reporting Foundation, which has been part of the IFRS Foundation since August 2022) published the International Integrated Reporting Framework (IIRF), characterised by a principle-based approach, to ensure an appropriate balance between content compliance requirements, the comparability and flexibility of the reports, information reliability and the ability to adapt to the specific nature of firms and circumstances.

The flip side of the coin is that this principle-based nature allows managers to interpret guidelines when preparing their integrated reports (Gerwanski *et al.*, 2019). Indeed, the concept of IRQ does not imply rigidly applying a framework (Pistoni *et al.*, 2018) but rather effectively explaining firms' ability to create value, considering the capital they use and affect (IIRC, 2021). Pistoni *et al.* (2018) referred to IRQ as the capacity of a report to convey strategic elements meaningfully, reflecting firm performance and value creation, while Vitolla *et al.* (2019) referred to disclosure quality as an assessment of the reliability, comparability and usefulness of the information presented by the reporting entity. Hence, an increase in information provided in corporate reports does not always imply an improvement in disclosure quality (Minutiello and Tettamanzi, 2022).

No single IRQ assessment method has yet been reached. Due to the lack of a unique checklist or coding framework to rank quality, scholars have operationalised IRQ using heterogeneous measures (Velte and Stawinoga, 2017). Some academics have developed their

Meditari Accountancy Research

MEDAR 32,7

200

scores, while others have used those provided by investment and accounting firms or proposed by scholars in their previous works.

Since the development of the IIRF, many authors have adopted self-constructed disclosure indices to proxy for IRQ (Lee and Yeo, 2016; Pavlopoulos *et al.*, 2017; Buallay *et al.*, 2021; Maama and Marimuthu, 2022; Senani *et al.*, 2022), obtained through a content analysis of integrated reports and a coding procedure based on a checklist. However, a self-constructed index may have certain limitations. Firstly, methodological issues arise concerning operationalising these scores based on the quantity of information disclosed in compliance with the framework (Kannenberg and Schreck, 2019). Subjective considerations and judgments can also influence quality assessments during the coding process and the creation of disclosure scores (Hossain *et al.*, 2023). Finally, the absence of theoretical guidance when making an aggregate IR score leads to assigning equal weights to each content element defined by the IIRF (Lee and Yeo, 2016).

Other authors have adopted the ratings on the quality of integrated reports launched by Ernst and Young [1] (EY) in 2011 for the top 100 Johannesburg Stock Exchange (JSE) listed firms (Martinez, 2016a; Barth *et al.*, 2017; Soumillion, 2018; Omran *et al.*, 2021), targeted at assessing the state of IR in South Africa and advocating for best practices and higher standards in IR preparation. This comprehensive measure accounts for the presence or absence of certain items and captures quality variations in integrated reports. Despite the extensive experience and independence of the adjudicators and the definition of a common EY score sheet, this is still a subjective measure. Another drawback is the limited number of firms for which this score is available.

Alternatively, in some studies, IRQ proxies were taken from rating scores developed by rating agencies and databases, such as Thomson Reuters Eikon, ASSET4, RobecoSAM and Bloomberg (Arguelles *et al.*, 2015; Serafeim, 2015; Loprevite *et al.*, 2018; Obeng *et al.*, 2021; Chouaibi *et al.*, 2022; Wu and Zhou, 2022), based on financial and environmental, social and governance (ESG) performance in firms' reports. Although database-derived proxies are available for large samples (De Villiers *et al.*, 2017b), these scores tend to be opaque (Dragomir and Dumitru, 2023).

The quality assessment of integrated reports was also based on scoreboards developed in the literature. For example, Leukhardt *et al.* (2022), Raimo *et al.* (2022) and Vitolla *et al.* (2020) adopted the scoring model designed by Pistoni *et al.* (2018), which focuses on four elements: background, assurance and readability, content and form to proxy IRQ. Hoang *et al.* (2020) used Zhou *et al.*'s (2017) measure of the level of alignment of an integrated report with the IIRF to construct the total disclosure score. Anifowose *et al.* (2020) used the scoring scheme for the quality of integrated reporting capital disclosure proposed by Anifowose *et al.* (2017).

Consequently, the lack of a unique IRQ measurement may represent a source of heterogeneity in the empirical study results. Meta-analysis determines whether adopting different proxies for IRQ measurement, referred to as moderating factors in meta-regression models, influences the relationship with the consequences explored in the literature.

2.2 The effects of integrated reporting quality

Over the past decade, several empirical studies have examined IRQ in mandatory and voluntary settings. IR represents the last frontier in corporate reporting as it fosters the provision of more inclusive and understandable financial and non-financial information in an integrated manner to address the limitations of traditional corporate reporting models. This holistic representation provides a comprehensive overview of how a firm creates value, its business model, risks and opportunities, strategy and resource allocation and outlook.

Moreover, IR can also be seen as a means for organisations to coordinate their strategies and value creation processes to address sustainability challenges, thanks to its focus on innovation and risk management, which enables them to contribute to advancing the sustainable development (Stefanescu, 2022; Hamad *et al.*, 2023; Rizzato *et al.*, 2024).

Given the rise in the popularity of IR, the existing literature has first investigated the determinants of its adoption and the quality of the disclosure. Recently, the emphasis has turned to pinpointing various consequences of IR (Vitolla *et al.*, 2019; Senani *et al.*, 2022). However, the findings have been fragmented and inconclusive (Zhou *et al.*, 2017; Crous *et al.*, 2022; Hossain *et al.*, 2023), and it is impossible to draw an overall conclusion regarding the effects of IRQ.

Several measures and multiple contexts have been adopted and investigated to capture these effects, and the theoretical discussions revolve around various approaches, which, however, are not free of overlap (Kannenberg and Schreck, 2019; Abogazia *et al.*, 2022). These include voluntary disclosure, signalling, legitimacy, agency and stakeholder theories (Hossain *et al.*, 2023; Songini *et al.*, 2023).

Many empirical studies have investigated the effect of IRQ in terms of market reaction, mainly proxied using Tobin's Q to capture how the market values a firm's assets in relation to their carrying amount. In the extant literature, two competing views have emerged concerning this relationship (Lee and Yeo, 2016). The first posits that IR benefits shareholders by providing higher-quality information that reduces information processing costs for investors. According to signalling theory, owing to the gap in access to information between managers and shareholders (Spence, 1973, 2002), the signals conveyed to outsiders through integrated disclosure can mitigate information asymmetry, facilitate financing, enhance the firm's value (Baiman and Verrecchia, 1996) and enable companies to distinguish themselves from less transparent peers. Barth et al. (2017) highlighted a significant positive association between IRQ and firm value in South Africa, where IR is mandatory for JSElisted companies. This result aligns with the findings of Lee and Yeo (2016), who documented that cross-sectional variation in IRO, proxied by a self-constructed disclosure index, is positively related to firm valuation after IR in the same geographical context, meaning that the benefits associated with IR outweigh the costs and reduce information asymmetry. Furthermore, they found that this association is stronger for firms characterised by organisational complexity, suggesting that IR is affected by firms' specificities. Martinez (2016b), Pavlopoulos et al. (2019), Abogazia et al. (2022) and Sun et al. (2022) showed that IRQ positively affects a firm's market valuation. According to agency theory, IR helps mitigate information asymmetry between managers and investors and the uncertainty about long-term corporate performance by improving the availability of company information (Leukhardt et al., 2022; Permatasari and Tjahjadi, 2023), thus leading to value creation (Jensen and Meckling, 1976; Abogazia et al., 2022). Additionally, Senani et al. (2022), in an emerging market context (Sri Lanka), emphasised that firms' market performance is positively related to the level of IR disclosure. From the signalling theory perspective, capital market participants incorporate the signals conveyed through integrated reports into their decision-making process, alleviating information asymmetry. Similarly, Dev (2020) scrutinised a sample of 30 listed firms on the Dhaka Stock Exchange and demonstrated that IR practice positively impacts firm value; no conclusive evidence was obtained regarding its relationship with stock liquidity. Finally, Serafeim (2015) found that integrated reporting practices lead to a change in the investor base of firms; specifically, IRQ attracts more longterm-oriented shareholders.

However, the second view posits that disclosing information can be counterbalanced by the potential loss of competitive advantage by revealing salient proprietary information to Meditari Accountancy Research

competitors, which may lead firms to disclose less. Buallay et al. (2021) exhibited that IR MEDAR negatively affects market performance (Tobin's O) for Islamic banks in Gulf Cooperative 32.7 Council countries. In this context, the effect of IR is value-destroying rather than valueenhancing. Soriya and Rastogi (2022) found an insignificant impact of IRQ on firm value in India. This can be attributed to the costs of acquiring and preparing new information systems and revealing competitive information. Finally, Sriani and Agustia (2020) showed a nonsignificant relationship with information asymmetry (proxied by spread).

By combining multiple theoretical perspectives and empirical results, the disclosure of financial and non-financial information in an integrated manner is crucial to meet financial capital providers' and other stakeholders' expectations and to mitigate information asymmetries between internal and external parties by improving transparency and decreasing monitoring costs for investors. Following these arguments and the widely documented positive relationship between disclosure quality and market valuation, we hypothesise that:

H1. There is a positive relationship between IRQ and market valuation.

From the stakeholder theory perspective (Freeman, 1994), stakeholders reward companies that exhibit transparency through enhanced disclosure quality in the form of improved firm performance (Matemane and Wentzel, 2019). Indeed, IR encompasses the financial capital providers' interests and those of other stakeholders (Kannenberg and Schreck, 2019; Hossain et al., 2023). Soriya and Rastogi (2022) observed that the IR disclosure quality index positively affects the operational performance of companies listed on the National Stock Exchange in India, measured by return on assets (ROA). Similarly, Pavlopoulos et al. (2019) reported a statistically significant and positive relationship between IR disclosure quality and firm profitability (ROA), whether mandatory or voluntary. Sun et al. (2022) demonstrated a significant positive relationship between the multiple capitals disclosure quality and profitability for Chinese listed companies.

However, in the extant literature, firms' financial performance is not uniformly defined (Soriya and Rastogi, 2022). For instance, Chouaibi et al. (2022) referred to return on equity (ROE) as an indicator of Islamic banks' performance in the Middle East and North Africa region and documented a significant and positive impact of IRQ. According to legitimacy theory, firms are incentivised to disclose financial and non-financial information to gain and preserve their social "license to operate" (Deegan, 2002; Hahn and Kühnen, 2013; Permatasari and Tjahjadi, 2023); therefore, IRQ is perceived as an important means of legitimisation (Chouaibi et al., 2022).

Despite the hypothesised positive relationship, Matemane and Wentzel (2019) found no significant effect of IRQ on ROA, ROE or economic value added for South African banks listed on the JSE. Buallay et al. (2021) found that for conventional banks, IR negatively affects ROE and ROA; however, this result does not hold for Islamic banks, implying that the expectations of signalling theory are unmet. Similarly, Adegboyegun et al. (2020) reported a non-significant impact of IR on corporate performance proxied by profit after tax in Nigeria.

Consistent with stakeholder and legitimacy theories, firms disclose interconnected financial and non-financial information to secure legitimacy and meet stakeholders' informational needs. In addition, previous works argued that shareholders and other stakeholder groups reward firms that show transparency and accountability through highquality integrated reports. Therefore, we hypothesise that:

H2. There is a positive association between IRQ and firms' financial performance.

From the perspective of voluntary disclosure theory, supplementing financial information with non-financial information is considered value-relevant for capital markets and the broader economy because it helps mitigate information asymmetry (Verrecchia, 1983; Dye, 1985; Barth *et al.*, 2017) and reduces uncertainty in assessing corporate performance (Zhou *et al.*, 2017). This approach aligns with IR's inherent aim of improving "the quality of information available to providers of financial capital to enable a more efficient and productive allocation of capital" (IIRC, 2021, p. 2).

Previous studies examining the effects of IRQ supported a negative link with the cost of capital. Zhou *et al.* (2017), Vitolla *et al.* (2020) and Chouaibi *et al.* (2022) consistently documented the negative impact of IRQ on the cost of equity capital, while Raimo *et al.* (2022) focused on the other component of the cost of capital (cost of debt), demonstrating the benefits in terms of access to third-party financing resources due to higher-quality integrated reports preparation. As a corollary of agency theory (Jensen and Meckling, 1976), lenders reward firm transparency, thereby diminishing costly information asymmetry (Verrecchia, 1983) and enabling a more thorough assessment of borrowers' default risk (Raimo *et al.*, 2022).

However, Anifowose *et al.* (2020) found that the overall IR disclosure score has no significant impact on the cost of financing if industry- and country-specific effects are not controlled for. This result is consistent with Barth *et al.* (2017) and Martinez (2016b).

From this review, it is not easy to determine the net overall effect of IRQ on the cost of financing or the role played by different factors, such as study characteristics, on the magnitude and direction of this association. Therefore, we hypothesise that:

H3. There is a relationship between IRQ and the cost of capital.

In discussions on the potential benefits of IR, some scholars, consistent with voluntary disclosure theory, have argued that the quality of disclosure may enrich the information environment (proxied by analysts' earnings forecast accuracy) for capital markets, as it improves analysts' understanding of corporate prospects (Beyer *et al.*, 2010; Permatasari and Tjahjadi, 2023). Consistent with agency theory, providing financial and non-financial information can mitigate asymmetry between companies and outsiders and enhance corporate transparency. In the context of mandatory IR adoption, Zúñiga *et al.* (2020) found that the Sustainability Disclosure Transparency Index (a proxy for IRQ) is associated with lower analysts' earnings forecast errors because it provides more accurate forward-looking information for financial analysts, making it easier to assess the future performance and enabling more accurate decision-making. This result is aligned with Zhou *et al.*'s (2017) findings, meaning that IR can convey useful information to capital markets participants such as analysts due to its focus on forward-looking information.

However, in a voluntary context, Leukhardt *et al.* (2022) showed a non-significant connection between IRQ and analysts' earnings forecast accuracy, highlighting the lack of relevance of IRQ. As the authors argued, discrepancies in the effects of IR may be influenced by the geographical setting under analysis. Indeed, unlike in voluntary settings, in the South African context, all companies are mandated to adopt IR, thus leading to a stronger impact of improved integrated disclosure quality.

Collectively, considering the arguments from voluntary disclosure theory, agency theory and previous research findings, the fourth hypothesis of this meta-analysis is as follows:

H4. IRQ is associated with lower analysts' earnings forecast errors.

Meditari Accountancy Research

Some authors have explored additional effects of IRQ. In particular, these studies MEDAR acknowledge the role of IR as a governance tool (Jensen and Meckling, 1976), which can 32.7 curb opportunistic managerial decisions, such as earnings manipulation and tax evasion. Paylopoulos et al. (2017) and Obeng et al. (2021) examined the association between IRO and agency costs under the theoretical lens of agency theory. They found that firms providing high-quality IR information exhibit lower agency costs, as managers are less likely to engage in aggressive, opportunistic behaviours because of improved transparency and investors' enhanced monitoring ability. According to the IIRF, IR practice can shift managerial focus from short- to long-term goals and promote the incorporation of stakeholder interests (IIRC, 2021). Obeng et al. (2021) also highlighted the influence of a country's level of stakeholder orientation on the relationship between IRQ and agency costs, reflecting the degree to which the interests of non-shareholders are integrated into management concerns.

> Furthermore, previous literature has pointed out a relationship between information asymmetry and accounting manipulation and the link between discretion afforded by the IIRF in report preparation and misreporting practices. Hoang *et al.* (2020) provided evidence that in an IR-mandatory setting, adherence to the IIRF diminishes the possibility for managers to manipulate the content of financial statements. This suggests that IR enhances firms' transparency, delivers more comprehensive information and mitigates agency conflicts and managers' misbehaviour. In the same geographical context, Donkor et al. (2022) found that IRQ is negatively associated with firms' corporate tax avoidance (CTA) practices, underscoring the strategic role of this reporting approach and its contribution to enhanced corporate transparency.

> Wu and Zhou (2022) supported the findings of previous studies regarding the association between IRQ and accrual-based earnings management because of improved transparency; instead, concerning real earnings management, they discovered the unintended consequence that firms with high IR scores engage in the manipulation of real transactions.

> Despite these conflicting results, agency theory suggests that disclosure quality generally bridges the gap between managers' and shareholders' interests, reduces information asymmetries and encourages long-term orientation. Hence, the following relationship is predicted:

H5. IRQ is related to the adoption of aggressive earnings management techniques.

3. Methodology: a meta-analytical review

According to Glass (1976), Hunter et al. (1982) and Rosenthal (1991), a meta-analysis is a rigorous alternative to traditional narrative reviews for assessing the findings of prior empirical studies that address the same research question on a specific topic. In the literature, this technique represents a useful systematic statistical tool to reconcile mixed evidence from several studies and overcome the problem of reduced statistical power derived from studies that consider small samples (Khlif et al., 2015).

This meta-analytic review contributes to the accounting and non-financial reporting stream of research by reconciling the conflicting results of the multiple effects of IRQ and providing objective conclusions that may not be straightforward from narrative literature reviews. To achieve this purpose, the steps hereinafter are followed. Firstly, empirical studies for this meta-analysis were identified. Secondly, they were coded based on different measures of IRQ and the variables they impact. Then, for each of the selected papers, the effect sizes, which measure the association between the independent and dependent variables, and the mean effect size were calculated. Moreover, the heterogeneity in the effect size estimates was tested. Finally, meta-regression is used to examine the potential sources of heterogeneity in the effect sizes and assess their impact.

3.1 Search strategy and inclusion criteria

This literature review focuses on empirical studies investigating the various effects of IRQ. The procedure follows Kepes *et al.* (2013), Ringquist (2013) and Stanley *et al.* (2013). Firstly, it involves identifying empirical papers investigating the research question of interest and then making a judgement about their inclusion in the sample for this meta-analysis (Opare *et al.*, 2021).

We began with an exhaustive electronic search using keywords to identify papers that matched the research question. The search keywords, drawing on Hossain *et al.*'s (2023) literature review on IR, were "integrated reporting", "International Integrated Reporting", "International Integrated Reporting Council", "IIRC", "integrated thinking", "integrated report consequence", "integrated report effect", "integrated report extent", "integrated report benefit", "integrated report insight" and "integrated report affect". Asterisks indicate related terms. The electronic search was conducted using the following major databases: Google Scholar, ScienceDirect, EBSCO, Emerald, Taylor and Francis, Springer, Wiley Online, Scopus, Web of Science and SSRN.

To ensure that no further empirical studies consistent with the research question of interest were omitted, we scanned the references of literature reviews on IR that adopted different methodologies, such as those of Vitolla *et al.* (2019), Veltri and Silvestri (2020), Hossain *et al.* (2023) and Nwachukwu (2022). Next, following Opare *et al.* (2021), we manually went through the reference lists of the initially found papers to search for additional studies.

To decide whether the papers from the initial search were deemed eligible for inclusion in the meta-analysis sample, we first went through the title, the abstract, the methodological section and, subsequently, the full text to verify if they met the inclusion criteria. Drawing on Ahamed et al.'s (2023) meta-analysis, the following inclusion criteria were applied:

- The paper is not an earlier version of the papers already included in the sample;
- The paper reports the relevant statistics for the calculation of the effect size. Studies
 that provided only descriptive statistics, as well as quantitative research papers with
 methodologies, such as event studies, that did not report the relevant statistics, were
 excluded (Dragomir and Dumitru, 2023). Even more so, conceptual papers and
 contributions using qualitative methodologies (case studies, interviews, etc.) were
 excluded (Dragomir and Dumitru, 2023);
- The studies include IRQ and its components as independent variables. Papers that investigated IR adoption only as an indicator variable were excluded;
- The papers are written only in English.

Papers not meeting these criteria were excluded once the full text had been analysed. For example, the paper by Cosma *et al.* (2018) was excluded because an event study methodology is applied, and the relevant statistics necessary for the calculation of the effect size are not reported. Moloi and Iredele's (2020) paper was excluded because only descriptive statistics and the independent sample *t*-test are used to analyse the data. Additionally, the papers by Wahl *et al.* (2020) and Botha *et al.* (2022) were excluded because they investigate the effects of preparing an integrated report, measured as a dummy variable, without assessing disclosure quality. Salvi *et al.*'s (2020) study was not included in the sample because it considers only the quality of disclosure of intangible forms of capital

Meditari Accountancy Research MEDAR within the IIRF. Finally, Conway's paper (Conway, 2019) was excluded because IRQ is used as a control variable in the multivariate regression model.

This process led to a final sample of 45 papers for this literature review from 2013 to 2022.

Table 1 shows, in alphabetical order according to the surname of the lead author, the sample of studies included in the meta-analysis and a detailed summary of their characteristics, namely, the year of publication, journal, country, mode of IR adoption, sample period and sample size and the sources of information that are necessary for the calculation of the effect sizes. Regarding the geographical settings covered by the studies, 26 examine a single country, particularly South Africa, whereas 19 are based on samples from multiple countries.

Table 2 reports the journal rankings issued by ABS, the UK Association of Business Schools and ABDC, the Australian Business Deans Council, in which the primary articles were published, and the number of studies obtained from each of them. The ranking categories are 4*, 4, 3, 2 and 1, and A*, A, B and C, where 4* and A* indicate the highest quality.

Most primary studies provide more than one effect size. Empirical papers conduct, for example, sensitivity analyses using different variables for the same construct, estimate different regressions using a sub-sample of firms, adopt alternative ways of measuring variables and rely on different control variables and model specifications (Wang and Shailer, 2018). The treatment of multiple effect sizes within a single primary study is a long-debated issue among meta-analysts (Hansen *et al.*, 2022). Effect size multiplicity allows for taking into account increased heterogeneity. Measurement choices, control variables and other study characteristics are likely to be sources of heterogeneity that affect the strength of the relationship object of interest, which would not be considered by traditional approaches based on the calculation of an average or median of these effect sizes (Hunter and Schmidt, 1990; Lipsey and Wilson, 2001). However, effect size multiplicity may violate the assumption of independence of observations (López-López *et al.*, 2018; Cheung, 2019). To address the issue of multiplicity, drawing on what has been argued by Opare *et al.* (2021), each effect size is reported separately (Dalton *et al.*, 2003; Carney *et al.*, 2011).

To check the internal consistency of the coding process, we computed Cronbach's alpha, which is above the minimum acceptable level of 0.7 (Hair *et al.*, 2019).

3.2 Computation of the effect sizes

To apply the meta-analytic technique, it is necessary to calculate the effect sizes, which indicates the magnitude and direction of the association between the independent (IRQ) and the dependent variables (Lipsey and Wilson, 2001).

Different methods of calculating effect sizes were adopted depending on the information reported in the multivariate analysis in the primary studies. Where the product-moment correlation coefficient r is reported, it was used to measure the effect size (Opare *et al.*, 2021). Whether r is not available, it was computed as a function of the t-statistic, z-statistic, standard error or p-value [2] using the formulas detailed in Table 3. However, not all studies directly report t-values. In this case, the effect size was obtained by first calculating t from the regression coefficients and standard errors using the following formula (Ahamed *et al.*, 2023):

$$t = \frac{b}{s} \tag{1}$$

Author (Year)	Journal	Country*	Adoption	Sample period	Sample size	Source of information
Abogazia <i>et al.</i> (2022)	JFRA	Egypt	Voluntary	2017-2020	200	Tables 4, 5, 6, 7
Adegboyegun <i>et al.</i> (2020)	CBM	Nigeria	Voluntary	2009-2018	96	Table 1
Affan (2019)	JEMA	Indonesia	Voluntary	2017	58	Table 2
Agarwal (2020)	CAFR	South Africa	Mandatory	2011-2019	1,705	Tables 4, 5, 6, 7
Anifowose et al. (2020)	ARA	Multiple	Mandatory/voluntary	2015-2018	13,944	Tables 3, 4
Arguelles et al. (2015)	UP	Multiple (39)	Mandatory/voluntary	2009–2013	1,920	Tables 5, 6
Barth <i>et al.</i> (2017)	AOS	South Africa	Mandatory	2011–2014	292	Tables 4, 5, 6, 7, 8, 9
Barth <i>et al.</i> (2022)	UP	South Africa	Mandatory	2011–2017	651	Tables 4, 5, 6, 7, 8
Buallay <i>et al.</i> (2021)	JIMA	Multiple (6)	Voluntary	2012-2016	190	Tables 8, 9
Caglio et al. (2020)	EAR	South Africa	Mandatory	2011-2016	444	Tables 4, 5, 6, 7, 8, OA3
Chanatup <i>et al.</i> (2020)	ESI	Thailand	Voluntary	2015	240	Table 2
Chouaibi <i>et al.</i> (2022)	JGR	Multiple (15)	Voluntary	2015-2020	402	Tables 5, 6
Ciubotariu <i>et al.</i> (2021)	JBEM	Multiple	Voluntary	2015-2016	45	Tables 1, 2
Cooray et al. (2020)	SUST	Sri Lanka	Voluntary	2015-2017	117	Tables 8, 9
Dey (2020)	IJDG	Bangladesh	Voluntary	2013-2018	144	Tables 5, 6
Donkor et al. (2022)	SAMPJ	South Africa	Mandatory	2011–2017	471	Tables 3, 4, 5, 6, 7
Eugster and Wagner (2020)	JIAAT	Switzerland	Voluntary	1999–2012	1,780	Tables 4, 5, 6, 7, 8
Fernando <i>et al.</i> (2017)	UP	Multiple	Voluntary	n.a.	195	Table 2
Hamad <i>et al.</i> (2022)	SAMPJ	Malaysia	Voluntary	2016-2020	332	Table 6
Hoang <i>et al.</i> (2020)	JAAR	South Africa	Mandatory	2011–2016	743	Tables 2, 3, 4, 5
Kim et al. (2017)	UP	Multiple (18)	Voluntary	2014-2015	138	Tables 4, 5
Kosovic and Patel (2013)	UP	South Africa	Mandatory/voluntary	2009, 2011	142	Tables 5-7, 5-8
Lee and Yeo (2016)	RQFA	South Africa	Mandatory	2010-2013	822	Tables 3, 4, 6, 7
Leukhardt <i>et al.</i> (2022)	CSREM	Multiple	Voluntary	2015-2016	202	Table 4
Loprevite et al. (2018)	SUST	Multiple	Mandatory/voluntary	2012-2016	1,815	Table 12
Maama and Marimuthu (2022)	JAAR	Multiple (10)	Mandatory/voluntary	2009–2018	1,338	Table 3
Martinez (2016)	UP	South Africa	Mandatory	2013-2015	192	Table 2
Martinez (2016)	UP	Multiple (15)	Voluntary	2011, 2015	192	Tables 6, 7, 8, 9
Matemane and Wentzel (2019)	BBS	South Africa	Mandatory/voluntary	2005-2014	35	Table 3
Obeng et al. (2021)	EAR	Multiple (35)	Voluntary	2009-2017	14,421	Tables 3, 4, 5, 7, 8, 9
Omran et al. (2021)	CSREM	South Africa	Mandatory	2014-2018	374	Tables 7, 8
						(continued)
						Aco
						M cour Re
						edi ntai sea 2
						tar ncy rcl
						i V 1

Table 1. Sample of studies included in the meta-analysis

JournalCountryJournalCountryJournalCountryJARMultiplJARRMultiplJARRMultiplJARRSri LantJARRSri LantJARRSci LantJARRNultiplJARRSci LantJARRNultiplJARRSci LantJARRMultiplOHELIYON	y* Adoption le (25) Voluntary ole (25) Voluntary ole (15) Voluntary he Voluntary ole Voluntary he Voluntary oluntary he Voluntary	Sample period Sample period 2011–2015 2011–2019 2017–2019 2017–2019 2017–2010 2017–2020 2017–2020 2017–2010	Sample size 382 307 399 133 5,726 93 108 94	Source of infor Tables 6, 8, 10, Tables 5, 6, 7, 9 Tables 5, 6, 7, 9 Tables 7, 9 Tables 7, 9 Tables 5, 6 Tables 5, 6 Tables 5, 4, 5, 12, 12, 12, 12, 12, 12, 12, 12, 12, 12
0) HELIYON Multipl	le Voluntary	2017		94
FP China	Voluntary	2012–2016		1,087
BSE Multipi	ie voluntary	2016	116	
MAR Malaysi	sia Voluntary	2012–2015	184	
CGIR Multipl	ale (43) Mandatory/vo	untary 2008–2015	19,926	
ABACUS South A	Africa Mandatory	2009–2012	443	
ARJ South A	Africa Mandatory	2013-2015	333	

Table 2. Dis	tribution of papers per journal and journal rankings				Meditari
Abbreviation	Journal	No. of studies	ABS ranking	ABDC ranking	Accountancy Research
AOS	Accounting, Organizations and Society	1	4*	A*	
EAR	European Accounting Review	2	3	A*	
ABACUS	Abacus	1	3	А	200
BSE	Business Strategy and the Environment	1	3	А	209
CGIR	Corporate Governance: An International Review	1	3	А	
CAFR	China Accounting and Finance Review	1	n.a.	А	
JACF	Journal of Applied Corporate Finance	1	1	А	
JIAAT	Journal of International Accounting, Auditing and Taxation	1	3	В	
RQFA	Review of Quantitative Finance and Accounting	1	3	В	
ARJ	Accounting Research Journal	1	2	В	
ARA	Asian Review of Accounting	1	2	В	
IJDG	International Journal of Disclosure and Governance	1	2	В	
JAAR	Journal of Applied Accounting Research	4	2	В	
JBEM	Journal of Business Economics and Management	1	2	В	
JMFM	Journal of Multinational Financial Management	1	2	В	
RIBAF	Research in International Business and Finance	1	2	В	
SAMPJ	Sustainability Accounting, Management and Policy Journal	2	2	В	
JIMA	Journal of Islamic Marketing	1	n.a.	В	
JGR	Journal of Global Responsibility	1	1	С	
CSREM	Corporate Social Responsibility and Environmental Management	2	1	С	
JFRA	Journal of Financial Reporting and Accounting	2	1	С	
BBS	Banks and Bank Systems	1	n.a.	С	
MAR	Management and Accounting Review	1	n.a.	С	
CBM	Cogent Business and Management	1	1	n.a.	
FP	Frontiers in Psychology	1	1	n.a.	
ESI	Entrepreneurship and Sustainability Issues	1	n.a.	n.a.	
HELIYON	Heliyon	1	n.a.	n.a.	
JEMA	Jurnal Ilmiah Bidang Akuntansi dan Manajemen	1	n.a.	n.a.	
SUST	Sustainability	2	n.a.	n.a.	
UP	Unpublished paper ^a	8	n.a.	n.a.	

Notes: n.a. indicates that the journal is not ranked; ^aincluding two master's theses, like in the meta-analysis by Ahamed et al. (2023) Source: Table by authors

where b is the coefficient estimate and s is the standard error, and then by applying the formula used for *t*-statistic availability.

Nevertheless, some undesirable statistical properties [3] are derived from r. Hence, to account for these problems and eliminate these biases, Lipsey and Wilson (2001) advocated transforming all estimated effect sizes into standard normal and variance-stabilising metrics using Fisher's Z transformation.

3.3 Estimation of the weighted mean effect size and standard error

Another key aspect is the selection of the fixed- or random-effects models. A random-effects model was chosen because it acknowledges that the primary studies are heterogeneous, even if they share the same relationship object of interest and these variations are derived from multiple sources, such as differences in the period and countries considered (Opare et al., 2021); thus, it would be unrealistic to presume that they share exactly the same true effect

32,7			
	Name	Formula	Description
	<i>Fixed-effects model</i> Effect size using correlation	$ES_r = r$	ES is the effect size and <i>r</i> is the product- moment correlation coefficient
210	Effect size using t-statistic	$ES_r = \sqrt{\frac{t^2}{(t^2 + df)}}$	ES obtained as a function of the <i>t</i> -statistic (<i>t</i>) and <i>df</i> that is the degrees of freedom given by n-1 where <i>n</i> is the sample size
	Effect size using z-statistic	$ES_r = \sqrt{\frac{Z^2}{N}}$	ES obtained as a function of the <i>z</i> -statistic (<i>Z</i>) and the total sample size (<i>N</i>)
	Z-transform effect size	$Z_r = 0.5 * log_e \left(\frac{1 + ES_r}{1 - ES_r}\right)$	Fisher's transformed effect size
	Standard error	$SE_{Z_r} = \frac{1}{\sqrt{n_i - 3}}$	Standard error of the transformed effect sizes
	Inverse variance	$w_i = \frac{1}{SE_{Z_r}^2}$	Weight associated with the sample size of each study
	Weighted mean effect size	$\overline{ES} = \frac{\sum (w_i ES_i)}{\sum w_i}$	Mean effect size calculated for all effect sizes in the meta-analysis
	Standard error of the mean	$SE_{\overline{ES}} = \frac{1}{\sqrt{\sum w_i}}$	Standard error of the mean effect size
	<i>Random-effects model</i> Chi-square statistic	$Q = \sum w_i \left(ES_i - \overline{ES} \right)^2$	<i>Q</i> -statistic for the test for heterogeneity, which is distributed as a chi-square with $k-1$ degrees of freedom. <i>K</i> is the number of effect sizes in
	Tau squared	$\tau^2 = \frac{Q - df}{\sum w_i - \frac{\sum w_i^2}{\sum w_i}}$	the study Between-study variance, where Q is the Q - statistic and df is the degrees of freedom
	Weight	$w_i^* = \frac{1}{v_i^*}$	Weight assigned to each study
	Total variance	$v_i^* = v_i + \tau^2$	Total variance obtained as the sum of within- study variance for study <i>i</i> and the between-
	Weighted mean effect size	$\overline{ES}^* = \frac{\sum (w_i^* ES_i)}{\sum w_i^*}$	studies variance (τ^2) Mean effect size in a random-effects model
	Variance of mean effect size	$v^* = \frac{1}{\sum w_i^*}$	Variance of mean effect size determined as the reciprocal of the sum of the weights
	Standard error of mean	$SE_{\overline{ES}^*} = \sqrt{v^*}$	Standard error of mean effect size
	circet size		(continued)

Table 3. Continued			Meditari
Name	Formula	Description	Accountancy Research
Lower limit	$\overline{ES}_i^* = \overline{ES}^* - 1.96 \left(SE_{\overline{ES}^*}\right)$	Lower limit of the confidence interval for the mean effect size	
Upper limit	$\overline{ES}_{i}^{*} = \overline{ES}^{*} + 1.96\left(SE_{\overline{ES}^{*}}\right)$ $\left \overline{ES}^{*}\right $	Upper limit of the confidence interval for the mean effect size	211
z-statistic	$Z^* = \frac{ }{SE_{\overline{ES}^*}}$	Z-value for the test of the significance of the mean effect size	
Fail-safe number	$X = \frac{k \left[k * (Z^*)^2 - 2,706\right]}{2,706}$	Number of studies that would be required to overturn significant results into insignificant. <i>k</i> is the number of studies in the meta-analysis and Z* is the z-statistic	

Sources: Lipsey and Wilson (2001), Opare et al. (2021) and Ahamed et al. (2023)

size (fixed-effects model). Given that the selected papers for the meta-analysis are not "identical", the weighted mean effect size and standard errors were calculated based on a random-effects model (Borenstein *et al.*, 2007), assigning different weights to different studies.

In the random-effects model, the variance of the effect sizes is split into its within-study (v_i) and between-study (τ^2) (unknown) components, which are used to determine the weights assigned to each study (Borenstein *et al.*, 2007).

Publication bias problem was also considered to check the robustness of the metaanalysis. This phenomenon, also called the "file drawer problem", is being investigated as it may seriously threaten the findings (Mokhtar, 2017). This problem stems from academics publishing more statistically significant and fewer insignificant study findings (Hansen *et al.*, 2022).

All the formulas are reported in Table 3.

3.4 Investigation of the sources of heterogeneity in the effect sizes

Heterogeneity in a meta-analysis can be defined as "the variability or differences between studies in the estimates of effects" (Song, 1999). Differences in study characteristics, such as measurement choices, control variables and other research specificities (detailed below), may lead to different findings and thus may represent sources of heterogeneity in the calculation of the effect sizes (Wang and Shailer, 2018; Salehi *et al.*, 2019). Therefore, the variables included in the meta-regression model depended on the characteristics of the primary studies from which the effect size estimates were derived.

Table 4 reports all the variables included in the meta-regressions and briefly describes them.

3.4.1 Research settings. The studies selected for the present meta-analysis relied on samples of organisations from several countries. Different geographical settings may affect the relationship between IRQ and specific outcome variables of interest. Whether preparing an integrated report is voluntary in most countries, South Africa represents an exception. In 2010, under the King III report, it became the first country worldwide to mandate the preparation of IR based on an "apply or explain" approach (Hossain *et al.*, 2023).

MEDAR	Table 4. Summary of the variables included in the meta-regression model			
32,7	Name of the variable	Description		
	Dependent variable Z _r	Fisher's Z-transformed effect size		
212	Independent variables Research settings Mandatory adoption	Dummy variable that takes the value of 1 if the primary studies examine geographical contexts in which IR adoption is mandatory and 0 in case of voluntary adoption		
	Measurement of IRQ consequences Market reaction			
	Tobin's Q	Dummy variable that takes the value of 1 if the effect size estimate is based on Tobin's Q as a measure of market reaction		
	Market share	Dummy variable that takes the value of 1 if the effect size estimate is based on market share as a measure of market reaction		
	Revenue growth rate	Dummy variable that takes the value of 1 if the effect size estimate is based on revenue growth rate as a measure of market reaction		
	Bid-ask spread	Dummy variable that takes the value of 1 if the effect size estimate is based on bid ask spread as a measure of market reaction		
	Expected future cash flows	Dummy variable that takes the value of 1 if the effect size estimate is based on expected future cash flows as a measure of market reaction		
	Market value of equity	Dummy variable that takes the value of 1 if the effect size estimate is based on market value of equity as a measure of market reaction		
	Abnormal return	Dummy variable that takes the value of 1 if the effect size estimate is based on abnormal return as a measure of market reaction		
	Long term investors	Dummy variable that takes the value of 1 if the effect size estimate is based on long term investors as a measure of market reaction		
	Stock price synchronicity	Dummy variable that takes the value of 1 if the effect size estimate is based on stock price synchronicity as a measure of market reaction		
	ESG controversies and assurance	Dummy variable that takes the value of 1 if the effect size estimate is based on ESG controversies and assurance as a measure of market reaction		
	Financial performance			
	ROA	Dummy variable that takes the value of 1 if the effect size estimate is based on ROA as a measure of financial performance		
	ROE	Dummy variable that takes the value of 1 if the effect size estimate is based on ROE as a measure of financial performance		
	Future economic value added	Dummy variable that takes the value of 1 if the effect size estimate is based on future economic value added as a measure of financial performance		
	Corporate environmental performance	Dummy variable that takes the value of 1 if the effect size estimate is based on corporate environmental performance as a measure of financial performance		
	Cost of capital	1		
	Cost of equity	Dummy variable that takes the value of 1 if the effect size estimate is based on cost of equity as a measure of cost of capital		
	Financial analysts' properties			
	Analyst forecast errors	Dummy variable that takes the value of 1 if the effect size estimate is based on analyst forecast errors as a measure of financial analysts' properties		
		(continued)		

Table 4. Continued

Name of the variable	Description	Accountancy
Analyst forecast dispersion	Dummy variable that takes the value of 1 if the effect size estimate is based on analyst forecast dispersion as a measure of financial analysts' properties	Research
Managerial decisions		040
Abnormal discretionary accruals	Dummy variable that takes the value of 1 if the effect size estimate is based on abnormal discretionary accruals as a measure of managerial decisions	213
Combined measure of real activities manipulation	Dummy variable that takes the value of 1 if the effect size estimate is based on combined measure of real activities manipulation as a measure of managerial decisions	
Agency costs	Dummy variable that takes the value of 1 if the effect size estimate is based on agency costs as a measure of managerial decisions	
Corporate tax avoidance	Dummy variable that takes the value of 1 if the effect size estimate is based on corporate tax avoidance as a measure of managerial decisions	
IR quality measurement		
Self-constructed disclosure index	Dummy variable that takes the value of 1 if the effect size estimate is based on a self-constructed disclosure index as a proxy for IRO	
EY's IRQ measure	Dummy variable that takes the value of 1 if the effect size estimate is based on EY's IRQ measure as a proxy for IRQ	
Score of rating agencies/database providers	Dummy variable that takes the value of 1 if the effect size estimate is based on a score provided by rating agencies/database providers as a proxy for IRO	
Scoreboards taken from the literature	Dummy variable that takes the value of 1 if the effect size estimate is based on a scoreboard developed in the literature as a proxy for IRO	
Control variables	IIIQ	
Year fixed effects	Dummy variable that takes the value of 1 if the primary study regression controls for year fixed effects	
Industry fixed effects	Dummy variable that takes the value of 1 if the primary study regression controls for industry fixed effects	
Country fixed effects	Dummy variable that takes the value of 1 if the primary study regression controls for country fixed effects	
Strength of results	0	
Year of publication	Year in which the paper is published or written, for those unpublished	
Publication status	Dummy variable that takes the value of 1 if the primary study is published in a journal	
Sample size	Logarithm of sample size of the effect size estimate	
Sample period	Number of years considered in the primary study	
Source: Table by authors		

Meditari

Approximately 41% of the primary regressions in the sample papers refer to mandatory IR adoption.

Therefore, we included a dummy variable in the meta-regressions to control for differences in disclosure environments.

3.4.2 *Measurement of integrated reporting quality and its consequences.* Meta-analytic accounting literature claims that the measurement of explanatory variables may affect the relationship under investigation (Khlif *et al.*, 2017).

MEDAR 32.7

214

A unique checklist or coding framework is unavailable to assess the quality of integrated report disclosures (Hossain *et al.*, 2023). In the empirical literature, different proxies for IRQ measurement have been used, such as self-constructed disclosure indices, Ernst and Young (EY) ratings on the quality of integrated reports, rating scores of rating agencies or database providers (ASSET4, Bloomberg, RobecoSAM) and scores derived from the scoreboards developed in the literature. Four dummy variables were included in the meta-regression to account for the potential effects related to the unavailability of a unique IRQ measure. They take the value of 1 if the primary study adopts that IRQ proxy and 0 otherwise.

Similarly, Vitolla *et al.* (2019) argued that IRQ can have multiple impacts. Hence, 21 variables (Table 4) clustered into five groups were considered to account for several outcome variables affected by IRQ. They take the value of 1 if the primary study investigates that effect and 0 otherwise. The variables are analysed separately in this review, as in the works of Opare *et al.* (2021) and Ahamed *et al.* (2023).

3.4.3 Control variables. The primary studies included in this meta-analysis controlled for the following effects: year, industry and country fixed effects.

3.4.4 Publication year. Another study characteristic that may moderate the relationship between IRQ and the outcome variables is the year of publication. This is because it is more likely that IRQ effects have been studied in recent years, owing to the gradual shift in focus from IR adoption to quality (Velte, 2022; Hossain *et al.*, 2023; Songini *et al.*, 2023). In addition, IRQ has improved over time owing to the progressive awareness of IR disclosures (Kannenberg and Schreck, 2019; Soriya and Rastogi, 2022).

To test whether the publication year has any relationship with the reported results, the year of publication or the year in which the paper is written (for those unpublished) was included in the meta-regression model (Opare *et al.*, 2021; Ahamed *et al.*, 2023). Since no time limitation was set during the article selection process, this variable ranges from 2013 to 2022.

3.4.5 Publication status. Publication bias is a significant issue in meta-analytic literature reviews (Stanley, 2005; Ahamed *et al.*, 2023). An important debate among scholars is whether unpublished papers should be considered together with published studies (Opare *et al.*, 2021). Some authors have argued that these papers had not been subjected to an indepth review process; thus, they decided to exclude unpublished studies from their meta-analyses (Hay *et al.*, 2006; Habib, 2012; Habib *et al.*, 2019). Others contended that the inclusion of unpublished studies mitigates the publication bias issue (Pomeroy and Thornton, 2008; Byron and Post, 2016; Lu and Taylor, 2016; Mokhtar, 2017; Schreder, 2018).

To mitigate the biases associated with journal publication, we included published and unpublished papers (Ahmed *et al.*, 2013; Opare *et al.*, 2021), such as conference proceedings (Afsay *et al.*, 2023) and master's theses (Ahamed *et al.*, 2023), and we tested through the meta-regressions whether publication status had a significant impact on effect sizes.

3.4.6 Sample size. As it gives weight to the effect sizes, the sample size is a crucial component of the meta-analysis (Ahamed *et al.*, 2023). The sample size of the studies analysed in this meta-analysis ranges from 35 to 19,926 firm-year observations. Studies with larger sample sizes, resulting in a smaller within-study variance, are assigned larger weights through the computation of their weight as the inverse of the total variance, which includes the within-study variance related to its sample size. This means that a study with a smaller variance receives a higher weight (Lipsey and Wilson, 2001).

3.4.7 Sample period. Finally, a variable to account for the period considered in the empirical studies object of analysis was included in the meta-regression model. According to Opare *et al.* (2021), articles considering a longer timeframe are expected to better represent the phenomena under investigation.

3.5 Meta-regression procedures

Meta-regression is a sophisticated tool that has not been widely used in accounting metaanalyses (Khlif *et al.*, 2017). It is an extension of subgroup meta-analysis, which aims to model and explore all sources of heterogeneity together (Khelil, 2023). According to Miller and Cardinal (1994), simultaneous analysis of (moderator) variables in multiple regressions enables an accurate evaluation of the relative explanatory power of each variable.

Considering the between-study heterogeneity, a random-effects meta-regression model was used. The between-study variability used in the expression to calculate the weights was estimated using the restricted maximum likelihood method. Knapp and Hartung's (2003) approach was adopted in the random-effects meta-regression to adjust the standard error estimates for the regression parameters. Following Opare *et al.* (2021) and Ahamed *et al.* (2023), we specified the model as follows:

$$Z_r = \beta_0 + \beta_1 EF + \beta_2 IRQ + \beta_3 RS + \beta_4 CV + \beta_5 SR + \mu, \mu \sim N(0, v_i + \tau^2)$$
(2)

where the dependent variable Z_r is the Fisher's transformed effect size for the effects of IRQ calculated from primary studies, and the independent variables account for the sources of heterogeneity identified in the previous paragraphs. Specifically, EF is a vector of dummy variables for the different outcome variables on which IRQ generates an effect (10 for market reaction, 4 for financial performance, 1 for cost of capital, 2 for financial analysts' properties and 4 for managerial decisions). Meta-regression analysis was applied, considering only variables with at least five effect sizes. IRQ is a vector of indicator variables for each different measure adopted to capture the quality of integrated reports; RS is a dummy variable representing the research settings in which IR adoption is mandatory or voluntary; CV is a vector of dummy variables related to the control variables included in the empirical articles; and SR is a vector of variables affecting the strength of results (year of publication, publication status, sample size and sample period). Table 4 summarised all these variables. v_i is the within-study variance and τ^2 is the between-study variance.

4. Results

This section discusses the results of the meta-analysis in detail.

4.1 Effect size results by primary studies

Table 5 presents the number of effect sizes collected from each primary empirical study included in the sample, listed in alphabetical order according to the surname of the lead author and summarises the effect size results for the impact of IRQ on several dependent variables. Hamad *et al.* (2022) reported the highest mean effect size (mean ES = 0.447). However, Pavlopoulos *et al.* (2017) reported the smallest mean effect size (mean ES = -0.695). The last row of Table 5 shows the overall number of effect sizes considered in this meta-analysis (653) and the mean effect size (0.006).

It clearly emerges a variation among the studies' effect sizes; the meta-regression model examines the factors that cause this heterogeneity.

The fail-safe number, which is the minimum number of studies necessary to reverse the results of the present meta-analysis, was computed using Rosenthal (1979) approach.

The fail-safe number is X = 2,554 (p = 0.0001), which far exceeds the tolerance level of 235 [4], meaning that the findings in Table 5 are not affected by the file drawer issue.

Meditari Accountancy Research

216	Abogazia <i>et al.</i> (2022) Adegboyegun <i>et al.</i> (2020) Affan (2019)	14	0.000		
216	Adegboyegun <i>et al.</i> (2020) Affan (2019)		0.306	0.259	0.353
216	Affan (2019)	2	0.010	-0.134	0.154
216		1	0.277	0.013	0.541
	Agarwal (2020)	7	0.041	0.023	0.059
	Anifowose <i>et al.</i> (2020)	18	0.050	0.046	0.054
	Arguelles <i>et al.</i> (2015)	30	0.032	0.018	0.047
	Barth et al. (2017)	51	0.005	-0.012	0.023
	Barth et al. (2022)	24	-0.078	-0.095	-0.061
	Buallay et al. (2021)	6	-0.061	-0.128	0.005
	Caglio et al. (2020)	65	-0.014	-0.026	-0.002
	Chanatup <i>et al.</i> (2020)	2	0.010	-0.080	0.100
	Chouaibi et al. (2022)	6	-0.142	-0.191	-0.093
	Ciubotariu et al. (2021)	12	0.440	0.353	0.527
	Cooray <i>et al.</i> (2020)	2	-0.057	-0.187	0.073
	Dey (2020)	6	0.008	-0.060	0.075
	Donkor et al. (2022)	38	-0.124	-0.139	-0.109
	Eugster and Wagner (2020)	131	0.053	0.048	0.057
	Fernando et al. (2017)	1	-0.014	-0.156	0.127
	Hamad <i>et al.</i> (2022)	3	0.447	0.385	0.510
	Hoang <i>et al.</i> (2020)	12	-0.079	-0.104	-0.054
	Kim et al. (2017)	5	-0.066	-0.158	0.026
	Kosovic and Patel (2013)	2	0.144	0.027	0.262
	Lee and Yeo (2016)	9	0.086	0.063	0.108
	Leukhardt <i>et al.</i> (2022)	1	-0.031	-0.170	0.108
	Loprevite <i>et al</i> . (2018)	2	0.081	0.041	0.122
	Maama and Marimuthu (2022)	7	-0.050	-0.070	-0.030
	Martinez (2016a)	1	-0.135	-0.277	0.008
	Martinez (2016b)	6	0.101	0.043	0.159
	Matemane and Wentzel (2019)	5	0.157	0.002	0.312
	Obeng <i>et al</i> . (2021)	23	-0.028	-0.033	-0.023
	Omran <i>et al</i> . (2021)	10	0.243	0.211	0.276
	Pavlopoulos et al. (2017)	8	-0.695	-0.731	-0.659
	Pavlopoulos et al. (2019)	12	0.101	0.068	0.134
	Raimo et al. (2022)	1	-0.154	-0.253	-0.056
	Senani <i>et al</i> . (2022)	4	-0.150	-0.236	-0.064
	Serafeim (2015)	18	0.040	0.032	0.048
	Soriya and Rastogi (2022)	4	0.014	-0.089	0.117
	Soumillion (2018)	3	0.024	-0.086	0.134
	Sriani and Agustia (2020)	14	0.000	-0.063	0.063
	Sun et al. (2022)	5	0.112	0.084	0.140
	Vitolla <i>et al</i> . (2020)	1	-0.193	-0.377	-0.008
	Wen <i>et al.</i> (2017)	16	-0.101	-0.138	-0.065
	Wu and Zhou (2022)	40	-0.020	-0.023	-0.017
	Zhou <i>et al.</i> (2017)	15	-0.071	-0.100	-0.042
	Zúñiga <i>et al</i> . (2020)	10	-0.280	-0.317	-0.243
					0.007

4.2 Test for heterogeneity

To investigate heterogeneity among the studies included in this meta-analysis, Cochran's Q (Cochran, 1954) and I^2 were examined. A significantly high *Q*-statistic indicates the presence of considerable effect size heterogeneity, whereas I^2 estimates the percentage of total variation across studies due to heterogeneity (Higgins *et al.*, 2003). Table 6 reports the results of the test for heterogeneity.

The overall *Q*-statistic of this meta-analysis is 13,804.17, and it is statistically significant at p < 0.001. This result strongly indicates the presence of between-study heterogeneity.

According to the categorisation proposed by Higgins *et al.* (2003), a value of I² equal to 95.28 implies high heterogeneity across studies, suggesting the need for a deeper comprehension of the factors underlying such variability. This means that approximately 95.28% of the observed variance among the studies is derived from real between-study differences and not from sampling errors (Ringquist, 2013).

4.3 Meta-regression results

A random-effects meta-regression model (equation (2)) was used to examine the various sources of heterogeneity in the sample of studies. To avoid too little statistical power of the results, following Dragomir and Dumitru (2023), we included in the model only dummies that account for the different choice of measures of IRQ consequences of at least five effect sizes. This model was estimated for each of the five clusters of outcome variable measures (market reaction, financial performance, cost of capital, financial analysts' properties and managerial decisions).

The meta-regression results are presented separately for each cluster of outcome variables affected by IRQ.

Market reaction. Table 7 presents the results of the meta-regression. The R-squared suggests that the variables included in the model explain 35.58% of the heterogeneity. The regression coefficients show that among the measures of the consequences of IRQ, only market share (coefficient = 0.3492139, p < 0.001) and bid-ask spread (coefficient = -0.0918198, p = 0.051) have a significant and, respectively, positive and negative impact on effect size relative to the mean impact of the choice of measure.

As regards IRQ assessment methods, they all have a negative and significant impact, except for EY's IRQ measure.

The control variable, mandatory adoption, has a negative and significant coefficient. This result indicates that the difference between the impact on the market reaction of geographical contexts in which IR adoption is mandatory and those in which it is voluntary is significant.

Description	Outcome
dfQP > Q% I2H2	652 13,804.17 < 0.001 95.28 21.17
Source: Table by authors	

Table 6. Summary of the results of the test for heterogeneity

Meditari Accountancy Research

MEDAR

32,7

218

Table 7. Random-effects meta-regression results for market reaction

Fisher Z	Coefficient	t-statistic
Tobin's Q	0.0059429	0.14
Market share	0.3492139***	4.38
Revenue growth rate	-0.0538305	-0.52
Bid-ask spread	-0.0918198*	-1.96
Expected future cash flows	0.0357809	0.63
Market value of equity	0.0504082	0.69
Abnormal return	-0.0288957	-0.40
Long term investors	0.0135467	0.13
Stock price synchronicity	-0.0270133	-0.31
ESG controversies and assurance	0.021327	0.29
Self-constructed disclosure index	-0.0955489*	-1.88
EY's IRQ measure	-0.0106241	-0.19
Score of rating agencies/database providers	-0.264849***	-4.10
Mandatory adoption	-0.1184693**	-2.57
Year fixed effects	-0.0691095*	-1.86
Industry fixed effects	-0.0354449	-0.99
Country fixed effects	0.0372029	0.42
Year of publication	-0.0102118	-0.78
Publication status	0.0885384	1.41
Sample size	0.0256406	0.77
Sample period	0.0074981	0.93
Constant	20.66327	0.78
Number of ES	291	
tau-squared (τ^2)	0.01916	
$I^{2}(\%)$	95.02%	
R-squared (%)	35.58%	

Notes: The dependent variable of the meta-regression is Fisher Z. Due to the issue of multicollinearity, the variable "Scoreboards taken from the literature" is omitted. ***p < 0.01; **p < 0.05; *p < 0.10 **Source:** Table by authors

Finally, among the control variables, only year fixed effects has a significant and negative coefficient. This means that, if included, these effects would have a negative impact on effect size.

Financial performance. Table 8 presents the results of the meta-regression. The R-squared suggests that the variables included in the model explain 16.85% of the heterogeneity. The regression coefficients show that among the measures of the consequences of IRQ, ROE (coefficient = 0.1012682, p = 0.083) and future economic value added (coefficient = 0.065253, p = 0.095) have a significant and positive impact on effect size relative to the mean impact of choice of measure.

Regarding IRQ assessment methods, none of the coefficients is significant.

The control variable, mandatory adoption, has a positive and significant coefficient, which shows the difference between the impact on financial performance of geographical contexts where IR adoption is mandatory and those where it is voluntary is significant.

The coefficients on control variables are not statistically significant.

The negative and significant coefficient of sample period indicates that longer sample periods have a negative impact on effect size.

Cost of capital. Table 9 summarises the results of the meta-regression for the cluster of measures related to the cost of capital. The R-squared suggests that the variables included in the model explain 23.66% of the heterogeneity. The regression coefficients show that among

			Accountancy
Fisher Z	Coefficient	t-statistic	Dessauch
	0.0502012	1.04	Research
ROA	0.0592912	1.04	
ROE	0.1012682*	1.75	
Future economic value added	0.065253*	1.68	
Corporate environmental performance	-0.0283162	-0.19	210
Self-constructed disclosure index	-0.1687835	-1.59	219
Score of rating agencies/database providers	-0.0153668	-0.10	
Mandatory adoption	0.2014818**	2.18	
Year fixed effects	-0.0311117	-0.57	
Industry fixed effects	0.0105856	0.19	
Year of publication	0.0265328	1.57	
Sample size	0.0005074	0.01	
Sample period	-0.0273364*	-1.77	
Constant	-53.34485	-1.56	
Number of ES	167		
tau-squared (τ^2)	0.02133		
I ² (%)	95.92%		
R-squared (%)	16.85%		

Notes: The dependent variable of the meta-regression is Fisher Z; Due to the issue of multicollinearity, the variables "EY's IRQ measure", "Scoreboards taken from the literature", "Country fixed effects" and "Publication status" are omitted. ***p < 0.01; **p < 0.05; *p < 0.10 **Source:** Table by authors

the measures of the consequences of IRQ, none have a significant impact on effect size relative to the mean impact of choice of measure; however, regarding IRQ assessment methods, only score of rating agencies/database providers has a negative and significant impact on effect size.

The results show that the differences between mandatory and voluntary IR adoption and the coefficients on the control variables are not statistically significant.

Among the variables that affect the strength of the results, only sample period has a significant positive coefficient, meaning that longer sample periods have a positive impact on effect size.

Financial analysts' properties. In Table 10, the R-squared suggests that the variables included in the model explain 85.69% of the heterogeneity. Among the measures of financial analysts' properties, none have a significant impact on effect size; however, regarding IRQ assessment methods, self-constructed disclosure index and score of rating agencies/database providers have a negative and significant impact on effect size relative to the mean impact of IRQ measures.

Only country fixed effects has a significant and positive coefficient, suggesting that the inclusion of these effects has a positive impact on effect size.

Finally, the results show that only sample size has a significant positive coefficient, indicating that using a larger sample size has a positive impact on effect size.

Managerial decisions. The R-squared presented in Table 11 suggests that the variables included in the model explain 50.94% of the heterogeneity. Among the measures of managerial decisions, only corporate tax avoidance has a significant and negative impact on effect size (coefficient = -0.3738053, p = 0.023). However, among the IRQ assessment methods, none have a significant coefficient.

MEDAR 32.7

52,7

220

Table 9. Random-effects meta-regression results for cost of capital

Fisher Z	Coefficient	t-statistic
Cost of equity	0.2283093	0.40
Self-constructed disclosure index	0.0476646	0.11
EY's IRQ measure	-0.3203937	-0.22
Score of rating agencies/database providers	-1.564073*	-1.97
Mandatory adoption	-2.276138	-1.28
Year fixed effects	0.5058432	0.37
Year of publication	-0.3506132	-0.68
Publication status	1.978684	0.83
Sample size	0.637508	0.76
Sample period	0.2559307**	2.31
Constant	704.1164	0.68
Number of ES	39	
tau-squared (τ^2)	0.4281	
I ² (%)	99.95%	
R-squared (%)	23.66%	

Notes: The dependent variable of the meta-regression is Fisher Z; Due to the issue of multicollinearity, the variables "Scoreboards taken from the literature", "Industry fixed effects" and "Country fixed effects" are omitted. ***p < 0.01; **p < 0.05; *p < 0.10**Source:** Table by authors

Table 10.	Random-effects m	eta-regression r	esults for fin	ancial analyst	s' properties
		···· · · · · · · · · ·			

Fisher Z	Coefficient	t-statistic
Analyst forecast errors	0.0383072	1.49
Self-constructed disclosure index	-0.0839463*	-1.71
Score of rating agencies/database providers	-0.10169**	-2.30
Year fixed effects	0.1017132	0.75
Industry fixed effects	-0.0746112	-0.75
Country fixed effects	0.2933906**	2.31
Sample size	0.4401172*	1.90
Constant	-1.185675*	-1.86
Number of ES	36	
tau-squared (τ^2)	0.000349	
$I^{2}(\%)$	10.43%	
R-squared (%)	85.69%	

Notes: The dependent variable of the meta-regression is Fisher Z; Due to the issue of multicollinearity, the variables "Analyst forecast dispersion", "EY's IRQ measure", "Scoreboards taken from the literature", "Mandatory adoption", "Year of publication", "Publication status" and "Sample period" are omitted. ***p < 0.01; **p < 0.05; *p < 0.10

Source: Table by authors

The results show that the differences between mandatory and voluntary IR adoption and the coefficients on control variables are not statistically significant.

Year of publication (or the year in which the paper is written, for those unpublished) has a significant and positive coefficient, which suggests that the recency of the publication (or writing) has a positive and significant impact on effect size.

Fisher Z	Coefficient	t-statistic	Accountancy
Abnormal discretionary accruals	-0.037225	-0.58	Research
Combined measure of real activities manipulation	-0.0218463	-0.34	
Agency costs	-0.0326388	-0.25	
Corporate tax avoidance	-0.3738053**	-2.31	004
Self-constructed disclosure index	0.0154594	0.16	221
EY's IRQ measure	-0.0034825	-0.03	
Score of rating agencies/database providers	0.0080131	0.07	
Mandatory adoption	0.1129867	0.62	
Year fixed effects	0.024783	0.21	
Industry fixed effects	-0.0343482	-0.21	
Country fixed effects	-0.1679498	-1.16	
Year of publication	0.1611457***	5.10	
Sample size	0.0199712	0.46	
Constant	-325.7531***	-5.12	
Number of ES	120		
tau-squared (τ^2)	0.02296		
$I^{2}(\%)$	99.15%		
R-squared (%)	50.94%		

Notes: The dependent variable of the meta-regression is Fisher *Z*; Due to the issue of multicollinearity, the variables "Scoreboards taken from the literature", "Publication status" and "Sample period" are omitted. ***p < 0.01; **p < 0.05; *p < 0.10**Source:** Table by authors

Table 12 summarises the results, which outlines the findings related to each hypothesis formulated in the meta-analysis.

4.4 Robustness test

To assess the robustness of the results obtained using the five different meta-regressions, five dummy variables were included to divide the set of measures into clusters and identify the effects of IRQ (market reaction, financial performance, cost of capital, financial analysts' properties and managerial decisions). Table 13 summarises these results.

The R-squared suggests that the variables included in the model explain 13,71% of the heterogeneity.

Among the choice of measures of IRQ effects, all except cost of capital have a positive and significant impact on effect size; adopting a different way of measuring the consequences of the disclosure quality impacts effect size. Regarding IRQ assessment methods, only EY's proxy for IRQ has a significant and positive impact (coefficient = 0.1480932, *t*-values = 3.13, *p* = 0.002).

Finally, the results for the research settings, control variables and measures indicating the strength of the results represent sources of heterogeneity that affect the results obtained in the primary empirical studies, consistent with what has been found in the separate meta-regressions.

5. Discussion

Answering Velte's (2019) call to conduct a meta-analysis on the subject of integrated reporting, this study reviews empirical papers dealing with IRQ effects, reconciling the

32,7	Hypotheses	Cluster	Main meta-regression results	Discussion	
222	H1	Market reaction	 Market share has a significant and positive impact on effect size relative to the mean impact of choice of measure Bid-ask spread has a significant and negative impact on effect size relative to the mean impact of choice of measure 	<i>H1</i> is supported for market share and bid-ask spread, an inverse measure for stock liquidity	
	H2	Financial performance	ROE and future economic value added have a significant and positive impact on effect size relative to the mean impact of choice of measure	<i>H2</i> is supported only for ROE and future economic value added	
	НЗ	Cost of capital	No significant impact on effect size	H3 is not supported	
	H4	Financial analysts' properties	No significant impact on effect size	H4 is not supported	
	Н5	Managerial decisions	Corporate tax avoidance has a significant and negative impact on effect size relative to the mean impact of choice of measure	<i>H5</i> is supported only for corporate tax avoidance practices	
	Source: Tabl	e by authors			

Table 12. Summary of the main meta-regression results.

MEDAR

conflicting results obtained in these studies and providing objective conclusions by adopting this technique.

Research on the effects of IRQ is expanding globally (Crous *et al.*, 2022; Hossain *et al.*, 2023). Many authors have investigated the impact of IRQ on market valuation using several proxies to capture how the market reacts to information in integrated reports. The results of our meta-analysis support the assumptions of agency theory, according to which increased disclosure quality reduces information asymmetry, fosters market confidence and ultimately leads to a positive influence on market valuations because the capital market channel reflects the quality of the information received (Barth *et al.*, 2017; Caglio *et al.*, 2020). Furthermore, the negative association between IRO and bid-ask spread, used as an inverse proxy for stock liquidity, suggests that high-quality integrated disclosure is associated with decreased information asymmetry (Leuz and Verrecchia, 2000; Barth et al., 2017, 2020). These results enrich the literature on the effect of IRO on market valuation by highlighting that companies that provide accurate and unbiased information to satisfy capital market participants' information needs can benefit from it. Indeed, by improving the quality of the information available and offering more comprehensive and reliable data about firms' value creation and value maintenance capabilities, companies can bridge the gap between themselves and their stakeholders, fostering trust and sustainable long-term relationships with investors and other stakeholders. Hence, the market rewards higher IRQ because it is assumed to reduce the opaqueness of information (De Villiers et al., 2017a).

The findings of the second meta-regression corroborate the hypothesised positive effect of IRQ on firms' financial performance measures, specifically on ROE and future economic value added. Confirming the assumption of stakeholder theory, when firms meet

Table 13. Robustness test	Meditari		
Fisher Z	Coefficient	t-statistic	Accountancy Research
Market reaction	0.2705603***	7.10	researen
Financial performance	0.2358665***	5.34	
Cost of capital	0.0834645	1.54	
Financial analysts' properties	0.2502129***	4.22	222
Self-constructed disclosure index	0.0367424	0.93	223
EY's IRQ measure	0.1480932***	3.13	
Score of rating agencies/database providers	-0.0484028	-0.98	
Mandatory adoption	-0.0748787**	-2.14	
Year fixed effects	-0.1119927***	-3.27	
Industry fixed effects	0.0227008	0.66	
Country fixed effects	-0.019086	-0.39	
Year of publication	0.0189352***	2.83	
Publication status	0.035439	0.98	
Sample size	0.0600507***	2.97	
Sample period	0.0237191***	3.84	
Constant	-38.73644***	-2.86	
Number of ES	653		
tau-squared (τ^2)	0.05351		
$I^{2}(\%)$	99.01%		
<i>R</i> -squared (%)	13.71%		

Notes: The dependent variable of the meta-regression is Fisher Z; Due to the issue of multicollinearity, the variables "Managerial decisions" and "Scoreboards taken from the literature" are omitted. ***p < 0.01; ** p < 0.05; * p < 0.10**Source:** Table by authors

stakeholders' informational needs, this is reflected in financial performance (Matemane and Wentzel, 2019); therefore, this review contributes to the stream of literature that supports the positive relationship between IRQ and firm financial performance, since IR contributes to improving transparency and mitigating information asymmetry (Abogazia *et al.*, 2022; Chouaibi *et al.*, 2022; Soriya and Rastogi, 2022). Providing greater information connectivity between governance, risks and opportunities, strategy, performance and prospects, IR considers all value-creation drivers and the stakeholders will reward firms demonstrating transparency and accountability (Chouaibi *et al.*, 2022).

The results do not support the hypothesised relationship between the quality of the integrated disclosure and the cost of capital and the association of IRQ with lower analyst earnings forecast errors because all meta-regression coefficients are not statistically significant. These findings challenge previous assumptions and highlight the need for further research to understand the interaction between IRQ and these variables.

Although some authors have highlighted the role of IR as a management tool that enables opportunistic behaviour (Wu and Zhou, 2022), the findings of our fifth meta-regression are consistent with the nature of IR: it exposes the firm's CTA schemes and hampers CTA strategies (Donkor *et al.*, 2022), suggesting that high-quality integrated reports can improve corporate transparency and encourage accountability and a trustworthy, honest and ethical corporate culture (Wu and Zhou, 2022). Disclosing the interactions among the six forms of capital can provide a more holistic representation of the information system, which supports both internal and external reporting and communication and fosters long-term orientation

(IIRC, 2021). Therefore, disclosing comprehensive information on corporate activities MEDAR through integrated reports can deter managers from engaging in opportunistic behaviours that may not align with shareholder interests or ethical standards, thereby addressing agency conflicts and manager misbehaviour (Hoang *et al.*, 2020).

6. Conclusion

From the analysis of 45 empirical studies from 2013 to 2022, this meta-analysis aims to contribute to the accounting literature and non-financial reporting stream of research by reconciling the mixed evidence obtained and providing objective conclusions to complement narrative literature reviews on IRO.

Several meta-analyses have been conducted in the accounting field, focusing on different subjects such as financial reporting, corporate governance, management accounting and auditing. Addressing Velte's (2019) suggestion to broaden the spectrum of this methodology to include IR, this is the first attempt to apply a meta-analysis to the effects of IRO, developing a better understanding and revealing the benefits and consequences of disclosure quality. In addition, the meta-regression model is a useful tool for controlling for factors that may affect the relationship of interest.

To capture different facets of the consequences of IRQ, they are operationalised in several ways and grouped according to the suggestions of Hossain *et al.* (2023). For market reaction, the meta-regression results highlight that among the measures of the IRQ consequences, the quality has a positive and significant impact on market share, while a negative one on bid-ask spread. In addition, all IRO assessment methods, except for EY's IRO, have a significant impact on the relationship relative to the mean level of impact on effect size. Among the research settings and the control variables included in the primary studies, only mandatory adoption and year fixed effect explain the variation in the effect size. For financial performance, the meta-regression confirms that ROE and future economic value added are positively and significantly influenced by the quality of the integrated disclosure. In addition, the research settings and sample periods considered in the empirical studies contribute to a better understanding of the differences in results. Conversely, regarding the cost of capital and financial analysts' property measures, none have a significant impact on effect size. Finally, the findings regarding managerial decisions show that IRO has a negative and significant impact only on corporate tax avoidance. For the control variables, only year of publication affects the relationship under investigation.

The findings of this literature review have several important implications for practitioners. They are particularly informative for non-financial report preparers, as they highlight the effects of high-quality reporting. Companies publishing integrated reports can benefit from it; high-quality integrated reports enhance firms' transparency and reduce information asymmetry with investors and other stakeholders. Through IR, companies can signal the capital market regarding their value-creation processes or risk status, which enhances firm valuation and improves corporate financial performance. In addition, preparing a high-quality integrated report highlights how integrated thinking shapes the corporate vision and aligns internal decision-making and actions with strategic goals and value creation (IFAC, 2020; Vitolla et al., 2020; IIRC, 2021, p. 3). Emphasising integrated thinking and a long-term orientation can promote a holistic understanding of business operations and the capital an organisation uses or affects. It can also support the path towards embedding the broader goal of contributing to sustainable development by providing the culture, strategy and tools needed to translate sustainable development targets into activities (Baboukardos et al., 2021; Stefanescu, 2022; De Villiers and Dimes, 2023; Hamad et al., 2023; Setia et al., 2024). The alignment of business approaches with the United Nations

32.7

Sustainable Development Goals (UN SDGs), driven by a shift in managers' focus from short-term targets to long-term value-creation processes at different organisational levels, can improve the understanding of how business activities affect sustainable development and steer investment flows to maximise value creation (Adams, 2017).

Given the multiple effects of IR in shifting managers' orientation, driving transformative change and enhancing accountability (Hossain *et al.*, 2023), regulators should evaluate promoting its adoption and setting more comprehensive and comprehensible reporting guidelines that can support the development of high-quality disclosure and curb managers' opportunistic behaviour (Hoang *et al.*, 2020; Hossain *et al.*, 2023; Permatasari *et al.*, 2023). In addition, in light of the recent evolution of the corporate sustainability reporting environment and the emergence of various frameworks and reporting solutions, IR's dual nature – as an external stakeholder reporting format and internal decision-making process through integrated thinking – could play a key role in meeting broader stakeholder informational needs and driving organisational change (Al-Htaybat and von Alberti-Alhtaybat, 2018; De Villiers and Dimes, 2023). Indeed, within the European Union (EU) context, the European Financial Reporting Advisory Group has advocated for an "integrated" approach to sustainability reporting and has embraced the key hallmarks of IR. Furthermore, IIRF's principle-based approach is compatible with International Sustainability Standards Board and Global Reporting Initiative reporting agendas (IIRC, 2021; De Villiers and Dimes, 2023).

However, preparers should acknowledge that IR will not unveil its effects unless it is perceived to be credible by stakeholders (Zhou *et al*, 2019; Hossain *et al.*, 2023). Independent external assurance of integrated reports can be used by managers to boost the reliability of the information provided, enhance the quality of corporate disclosures and mitigate information asymmetry (Maroun, 2019, 2022; Vitolla *et al.*, 2020; IIRC, 2021; Raimo *et al.*, 2022; Hossain *et al.*, 2023). Firms, especially those with a high level of integration of sustainability into their corporate strategy, are more willing to enhance transparency and reduce inaccuracies and inconsistencies in the reporting process through engaging in the assurance of integrated reports, which is positively valued by stakeholders (Caglio *et al.*, 2020; Baboukardos *et al.*, 2021).

As meta-analyses offer a useful snapshot of a controversial topic (Velte, 2019) and allow aggregation of conflicting results of prior studies, this review contributes to the accounting literature and non-financial reporting research stream by providing a broader overview of the effects of IRQ, which opens new possibilities for future research developments. Firstly, future studies should continue to investigate IRQ, focusing on its real effects, that is, how it affects internal decisions, given the twofold objective of this innovative corporate reporting tool: to expand the firm's transparency for financial capital providers and, at the same time, to foster a focus on the long-term value-creation process.

However, as highlighted by Hossain *et al.* (2023), prior works have mainly focused on the external effects of IRQ (e.g. market reaction) while overlooking its relationship with engagement in opportunistic practices, such as earnings management. Another little-explored area of research is the relationship with firms' ESG performance. Despite few authors have extended their analysis beyond the financial consequences of IRQ, exploring its impact on non-financial performance may be interesting, leveraging IR's dual nature to serve as an informational resource and a driver of transformation (Omran *et al.*, 2021).

Furthermore, meta-analysis allows an understanding of the sources of heterogeneity and reconciles conflicting evidence obtained by different scholars in their studies, thereby enhancing the generalisability of the findings. The results of our meta-regressions suggest that one source of heterogeneity stems from the multiple ways adopted in primary studies to operationalise the IRQ. Hence, future works should attempt to define a common framework for evaluating IRQ

Meditari Accountancy Research (Minutiello and Tettamanzi, 2022). Finally, future studies should explore the effects of IRQ by adopting different empirical designs, such as case studies, to gain deeper insights into the role of IR as a management tool and its impact on organisations' decision-making processes.

However, this study has some limitations related to the inherent characteristics of the adopted methodology. The validity of the selection procedure and the studies included in the analysis may be questioned (Alareeni, 2019). Unpublished papers are also included to mitigate publication bias (Opare *et al.*, 2021; Ahamed *et al.*, 2023), but access to them is not always straightforward. Including only empirical papers that report relevant statistics for the calculation of the effect sizes is another shortcoming of this methodology. Therefore, this study should be complemented by qualitative papers on this topic. Another limitation of this meta-analysis is the small sample size of the studies investigating the effects of IRQ. Additional papers to support the findings of the present work would be beneficial, paving the way for repeating this analysis in the future and focusing on other consequences that were not considered in this study.

Despite these limitations, this review provides useful insights into the effects of IRQ at different levels of analysis.

Notes

- 1. To foster the quality of integrated reports, EY divides them into five groups: "top 10", "excellent", "good", "average" and "progress to be made", except for 2011 when the category "average" was not used.
- 2. Drawing on Ahamed *et al.* (2023), the estimation of effect sizes based on *p*-value was obtained using a web-based effect size calculator developed by David B. Wilson of George Mason University (Wilson's book *Practical Meta-analysis*).
- 3. The main justification for using Fisher's Z transformation is that it removes three types of statistical problems: the fact that r underestimates the effect sizes, it is bounded between -1 and 1 and the dependence of its variance on the value of r itself (Lipsey and Wilson, 2001; Ringquist,

2013). About this last point, the variance for the untransformed *r* was computed as $V_r = \frac{(1-r^2)^2}{n-1}$ where *n* is the sample size, and the formula for the *z*-transformed effect size was $V_{zr} = \frac{1}{n-2}$.

4. According to Rosenthal (1979), the tolerance level was computed using the following formula: Y = 5 * k + 10, where k is the number of studies included in the present meta-analysis.

References

*Paper included in the meta-analysis

- *Abogazia, A.H., Hashim, H.A., Salleh, Z. and Ettish, A.A. (2022), "The moderating effect of external financing on the relationship between integrated reporting and firm value in Egypt", *Journal of Financial Reporting and Accounting*, doi: 10.1108/JFRA-05-2022-0195.
- Adams, C. (2017), "The sustainable development goals, integrated thinking and the integrated report", available at: https://integratedreporting.ifrs.org/wp-content/uploads/2017/09/SDGs-and-the-integrated-report_full17.pdf
- *Adegboyegun, A.E., Alade, M.E., Ben-Caleb, E., Ademola, A.O., Eluyela, D.F. and Oladipo, O.A. (2020), "Integrated reporting and corporate performance in Nigeria: evidence from the banking industry", *Cogent Business and Management*, Vol. 7 No. 1, p. 1736866, doi: 10.1080/ 23311975.2020.1736866.
- *Affan, M.W. (2019), "Integrated reporting and corporate performance: empirical evidence of the IIRC framework adoption", *JEMA: Jurnal Ilmiah Bidang Akuntansi Dan Manajemen*, Vol. 16 No. 2, pp. 181-186.

32.7

MEDAR

Afsay, A., Tahriri, A. and Rezaee, Z. (2023), "A meta-analysis of factors affecting acceptance of information technology in auditing", <i>International Journal of Accounting Information Systems</i> , Vol. 49, p. 100608, doi: 10.1016/j.accinf.2022.100608.	Meditari Accountancy
*Agarwal, N. (2020), "Integrated reporting and the informativeness of annual reports. Does textual coherence matter?", (accessed 30 December 2022).	Research
Ahamed, F.T., Houqe, M.N. and van Zijl, T. (2023), "Meta-analysis of the impact of financial constraints on firm performance", <i>Accounting and Finance</i> , Vol. 63 No. 2, pp. 1671-1707, doi: 10.1111/acfi.12923.	227
Ahmed, K., Chalmers, K. and Khlif, H. (2013), "A meta-analysis of IFRS adoption effects", <i>The</i> <i>International Journal of Accounting</i> , Vol. 48 No. 2, pp. 173-217, doi: 10.1016/j. intacc.2013.04.002.	
Al-Htaybat, K. and Von Alberti-Alhtaybat, L. (2018), "Integrated thinking leading to integrated reporting: case study insights from a global player", <i>Accounting, Auditing and Accountability</i> <i>Journal</i> , Vol. 31 No. 5, pp. 1435-1460, doi: 10.1108/AAAJ-08-2016-2680.	
Alareeni, B.A. (2019), "The associations between audit firm attributes and audit quality-specific indicators: a meta-analysis", <i>Managerial Auditing Journal</i> , Vol. 34 No. 1, pp. 6-43, doi: 10.1108/MAJ-05-2017-1559.	
*Anifowose, M., Abang, S. and Zakari, M.A. (2020), "Integrated capitals reporting and companies' sustainable value: evidence from the Asian continent", <i>Asian Review of Accounting</i> , Vol. 28 No. 4, pp. 567-589, doi: 10.1108/ARA-10-2019-0184.	
Anifowose, M., Abdul Rashid, H.M. and Annuar, H.A. (2017), "Intellectual capital disclosure and corporate market value: does board diversity matter?", <i>Journal of Accounting in Emerging</i> <i>Economies</i> , Vol. 7 No. 3, pp. 369-398, doi: 10.1108/JAEE-06-2015-0048.	
*Arguelles, M., Balatbat, M. and Green, W. (2015), <i>Is There an Early-Mover Market Value Effect for Signalling Adoption of Integrated Reporting</i> ?, working paper, University of New South Wales, Sydney, 25 September.	
Baboukardos, D., Mangena, M. and Ishola, A. (2021), "Integrated thinking and sustainability reporting assurance: International evidence", <i>Business Strategy and the Environment</i> , Vol. 30 No. 4, pp. 1580-1597, doi: 10.1002/bse.2695.	
Baiman, S. and Verrecchia, R.E. (1996), "The relation among capital markets, financial disclosure, production efficiency, and insider trading", <i>Journal of Accounting Research</i> , Vol. 34 No. 1, pp. 1-22, doi: 10.2307/2491329.	
*Barth, M.E., Cahan, S.F., Chen, L. and Venter, E.R. (2017), "The economic consequences associated with integrated report quality: capital market and real effects", <i>Accounting, Organizations and Society</i> , Vol. 62, pp. 43-64, doi: 10.1016/j.aos.2017.08.005.	
Barth, M.E., Cahan, S.F., Chen, L. and Venter, E.R. (2020), "Archival research informing the dual objective of integrated reporting", in de Villiers, C., Hsiao, P.C.K. and Maroun, W. (Eds), <i>The Routledge Handbook of Integrated Reporting</i> , Routledge, Abingdon, pp. 183-193.	
*Barth, M.E., Cahan, S.F., Chen, L. and Venter, E.R. (2022), "Firm-specific information and proprietary costs: evidence from mandated integrated reports", doi: 10.2139/ssrn.3857927. (accessed 26 July 2023).	
Beyer, A., Cohen, D.A., Lys, T.Z. and Walther, B.R. (2010), "The financial reporting environment: review of the recent literature", <i>Journal of Accounting and Economics</i> , Vol. 50 No. 2-3, pp. 296-343, doi: 10.1016/j.jacceco.2010.10.003.	
Borenstein, M., Hedges, L. and Rothstein, H. (2007), "Meta-analysis: fixed effect vs. random effects", available at: www.meta-analysis.com	
Botha, M., Oberholzer, M. and Middelberg, S.L. (2022), "Water governance disclosure: the role of integrated reporting in the food, beverage and tobacco industry", <i>Meditari Accountancy</i>	

Research, Vol. 30 No. 7, pp. 256-279, doi: 10.1108/MEDAR-09-2020-1006.

MEDAR 32,7	*Buallay, A., Al Hawaj, A.A. and Hamdan, A. (2021), "Integrated reporting and performance: a cross- country comparison of GCC Islamic and conventional banks", <i>Journal of Islamic Marketing</i> , Vol. 12 No. 8, pp. 1619-1636, doi: 10.1108/JIMA-08-2017-0084.
	Busco, C., Granà, F. and Achilli, G. (2021), "Understanding integrated thinking: evidence from the field, the development of a framework and avenues for future research", <i>Meditari Accountancy Research</i> , Vol. 29 No. 4, pp. 673-690, doi: 10.1108/MEDAR-04-2021-1263.
228	Byron, K. and Post, C. (2016), "Women on boards of directors and corporate social performance: a meta-analysis", <i>Corporate Governance: An International Review</i> , Vol. 24 No. 4, pp. 428-442, doi: 10.1111/corg.12165.
	*Caglio, A., Melloni, G. and Perego, P. (2020), "Informational content and assurance of textual disclosures: evidence on integrated reporting", <i>European Accounting Review</i> , Vol. 29 No. 1, pp. 55-83, doi: 10.1080/09638180.2019.1677486.
	Carney, M., Gedajlovic, E.R., Heugens, P., Van Essen, M. and Van Oosterhout, J. (2011), "Business group affiliation, performance, context, and strategy: a meta-analysis", <i>Academy of Management Journal</i> , Vol. 54 No. 3, pp. 437-460, doi: 10.5465/amj.2011.61967812.
	*Chanatup, S., Aujirapongpan, S. and Ritkaew, S. (2020), "The influence of corporate governance mechanism on the integrated financial reporting and investment risk of Thai listed companies", <i>Entrepreneurship and Sustainability Issues</i> , Vol. 7 No. 4, pp. 2818-2831, doi: 10.9770/ jesi.2020.7.4(16).
	Cheung, M.W.L. (2019), "A guide to conducting a meta-analysis with non-independent effect sizes", <i>Neuropsychology Review</i> , Vol. 29 No. 4, pp. 387-396, doi: 10.1007/s11065-019-09415-6.
	*Chouaibi, Y., Belhouchet, S., Chouaibi, S. and Chouaibi, J. (2022), "The integrated reporting quality, cost of equity and financial performance in Islamic banks", <i>Journal of Global Responsibility</i> , Vol. 13 No. 4, pp. 450-471, doi: 10.1108/JGR-11-2021-0099.
	*Ciubotariu, M.S., Socoliuc, M., Grosu, V., Mihailă, S. and Cosmulese, C.G. (2021), "Modeling the relationship between integrated reporting quality and sustainable business development", <i>Journal of Business Economics and Management</i> , Vol. 22 No. 6, pp. 1531-1550, doi: 10.3846/jbem.2021.15601.
	Cochran, W.G. (1954), "The combination of estimates from different experiments", <i>Biometrics</i> , Vol. 10 No. 1, pp. 101-129, doi: 10.2307/3001666.
	Conway, E. (2019), "Quantitative impacts of mandatory integrated reporting", <i>Journal of Financial Reporting and Accounting</i> , Vol. 17 No. 4, pp. 604-634, doi: 10.1108/JFRA-08-2018-0066.
	*Cooray, T., Senaratne, S., Gunarathne, A.D.N., Herath, R. and Samudrage, D. (2020), "Does integrated reporting enhance the value relevance of information? Evidence from Sri Lanka", <i>Sustainability</i> , Vol. 12 No. 19, p. 8183, doi: 10.3390/su12198183.
	Cosma, S., Soana, M.G. and Venturelli, A. (2018), "Does the market reward integrated report quality?", <i>African Journal of Business Management</i> , Vol. 12 No. 4, pp. 78-91, doi: 10.5897/ AJBM2017.8469.
	Crous, C., Battisti, E. and Leonidou, E. (2022), "Non-financial reporting and company financial performance: a systematic literature review and integrated framework", <i>EuroMed Journal of</i> <i>Business</i> , Vol. 17 No. 4, pp. 652-676, doi: 10.1108/EMJB-12-2020-0134.
	Dalton, D.R., Daily, C.M., Certo, S.T. and Roengpitya, R. (2003), "Meta-analyses of financial performance and equity: fusion or confusion?", <i>Academy of Management Journal</i> , Vol. 46 No. 1, pp. 13-26, doi: 10.5465/30040673.
	De Villiers, C., Hsiao, P.C.K. and Maroun, W. (2017a), "Developing a conceptual model of influences around integrated reporting, new insights and directions for future research", <i>Meditari</i> <i>Accountancy Research</i> , Vol. 25 No. 4, pp. 450-460, doi: 10.1108/MEDAR-07-2017-0183.
	De Villiers, C., Venter, E.R. and Hsiao, P.C.K. (2017b), "Integrated reporting: background, measurement issues, approaches and an agenda for future research", <i>Accounting and Finance</i> , Vol. 57 No. 4, pp. 937-959, doi: 10.1111/acfi.12246.

De Villiers, C. and Dimes, R. (2023), "Will the formation of the international sustainability standards board result in the death of integrated reporting?", <i>Journal of Accounting & Organizational</i> <i>Change</i> , Vol. 19 No. 2, pp. 279-295, doi: 10.1108/JAOC-05-2022-0084.	Meditari Accountancy Recoarch
Deegan, C. (2002), "Introduction: the legitimising effect of social and environmental disclosures–a theoretical foundation", <i>Accounting, Auditing and Accountability Journal</i> , Vol. 15 No. 3, pp. 282-311, doi: 10.1108/09513570210435852.	Research
Derfuss, K. (2009), "The relationship of budgetary participation and reliance on accounting performance measures with individual-level consequent variables: a meta-analysis", <i>European Accounting Review</i> , Vol. 18 No. 2, pp. 203-239, doi: 10.1080/09638180802652371.	229
*Dey, P.K. (2020), "Value relevance of integrated reporting: a study of the Bangladesh banking sector", International Journal of Disclosure and Governance, Vol. 17 No. 4, pp. 195-207, doi: 10.1057/ s41310-020-00084-z.	
*Donkor, A., Djajadikerta, H.G., Mat Roni, S. and Trireksani, T. (2022), "Integrated reporting quality and corporate tax avoidance practices in South Africa's listed companies", <i>Sustainability</i> <i>Accounting, Management and Policy Journal</i> , Vol. 13 No. 4, pp. 899-928, doi: 10.1108/SAMPJ- 03-2021-0116.	
Dragomir, V.D. and Dumitru, M. (2023), "Does corporate governance improve integrated reporting quality? A meta-analytical investigation", <i>Meditari Accountancy Research</i> , Vol. 31 No. 6, doi: 10.1108/MEDAR-03-2022-1618.	
Dye, R.A. (1985), "Disclosure of nonproprietary information", <i>Journal of Accounting Research</i> , Vol. 23 No. 1, pp. 123-145, doi: 10.2307/2490910.	
*Eugster, F. and Wagner, A.F. (2020), "Value reporting and firm performance", <i>Journal of International Accounting, Auditing and Taxation</i> , Vol. 40, pp. 1-18, doi: 10.1016/j.intaccaudtax.2020.100319.	
*Fernando, K., Dharmawati, R., Sriani, D., Shauki, E.R. and Diyanty, V. (2017), "Does integrated reporting approach enhance the value relevance of accounting information?: evidence from Asian firms", 6th International Accounting Conference (IAC 2017), Advances in Economics, Business and Management Research, pp. 112-117, doi: 10.2991/iac-17.2018.20.	
Freeman, R.E. (1994), "The politics of stakeholder theory: some future directions", <i>Business Ethics Quarterly</i> , Vol. 4 No. 4, pp. 409-421, doi: 10.2307/3857340.	
García-Meca, E. and Sánchez-Ballesta, J.P. (2010), "The association of board independence and ownership concentration with voluntary disclosure: a meta-analysis", <i>European Accounting</i> <i>Review</i> , Vol. 19 No. 3, pp. 603-627, doi: 10.1080/09638180.2010.496979.	
García-Sánchez, I. and Noguera-Gámez, L. (2017), "Integrated reporting and stakeholder engagement: the effect on information asymmetry", <i>Corporate Social Responsibility and Environmental</i> <i>Management</i> , Vol. 24 No. 5, pp. 395-413, doi: 10.1002/csr.1415.	
Gerwanski, J., Kordsachia, O. and Velte, P. (2019), "Determinants of materiality disclosure quality in integrated reporting: empirical evidence from an international setting", <i>Business Strategy and the Environment</i> , Vol. 28 No. 5, pp. 750-770, doi: 10.1002/bse.2278.	
Glass, G.V. (1976), "Primary, secondary, and meta-analysis of research", <i>Educational Researcher</i> , Vol. 5 No. 10, pp. 3-8, doi: 10.3102/0013189X005010003.	
Habib, A. (2012), "Non-audit service fees and financial reporting quality: a meta-analysis", <i>Abacus</i> , Vol. 48 No. 2, pp. 214-248, doi: 10.1111/j.1467-6281.2012.00363.x.	
Habib, A., Bhuiyan, M.B.U., Huang, H.J. and Miah, M.S. (2019), "Determinants of audit report lag: a meta-analysis", <i>International Journal of Auditing</i> , Vol. 23 No. 1, pp. 20-44, doi: 10.1111/ijau.12136.	
Hahn, R. and Kühnen, M. (2013), "Determinants of sustainability reporting: a review of results, trends, theory, and opportunities in an expanding field of research", <i>Journal of Cleaner Production</i> , Vol. 59, pp. 5-21, doi: 10.1016/j.jclepro.2013.07.005.	
Hair, J.F., Black, W.C., Babin, B.J. and Anderson, R.E. (2019), <i>Multivariate Data Analysis</i> , 8th ed., Cengage Learning, Boston.	

MEDAR 32,7	*Hamad, S., Lai, F.W., Shad, M.K., Khatib, S.F. and Ali, S.E.A. (2022), "Assessing the implementation of sustainable development goals: does integrated reporting matter?", <i>Sustainability Accounting,</i> <i>Management and Policy Journal</i> , Vol. 14 No. 1, doi: 10.1108/SAMPJ-01-2022-0029.
220	Hamad, S., Lai, F.W., Shad, M.K., Khatib, S.F.A. and Ali, S.E.A. (2023), "Assessing the implementation of sustainable development goals: does integrated reporting matter?", <i>Sustainability Accounting, Management and Policy Journal</i> , Vol. 14 No. 1, pp. 49-74, doi: 10.1108/SAMPJ-01-2022-0029.
230	Hansen, C., Steinmetz, H. and Block, J. (2022), "How to conduct a meta-analysis in eight steps: a practical guide", <i>Management Review Quarterly</i> , Vol. 72 No. 1, pp. 1-19, doi: 10.1007/s11301-021-00247-4.
	Hay, D.C., Knechel, W.R. and Wong, N. (2006), "Audit fees: a meta-analysis of the effect of supply and demand attributes", <i>Contemporary Accounting Research</i> , Vol. 23 No. 1, pp. 141-191, doi: 10.1506/4XR4-KT5V-E8CN-91GX.
	Higgins, J.P.T., Thompson, S.G., Deeks, J.J. and Altman, D.G. (2003), "Measuring inconsistency in meta-analyses", <i>BMJ</i> , Vol. 327 No. 7414, pp. 557-560, doi: 10.1136/bmj.327.7414.557.
	*Hoang, T.G., Vu, T.K., Nguyen, H.T. and Luu, H.N. (2020), "Mandatory integrated reporting disclosure and corporate misreporting", <i>Journal of Applied Accounting Research</i> , Vol. 21 No. 3, pp. 363-382, doi: 10.1108/JAAR-02-2019-0025.
	Hossain, A., Bose, S. and Shamsuddin, A. (2023), "Diffusion of integrated reporting, insights and potential avenues for future research", <i>Accounting and Finance</i> , Vol. 63 No. 2, pp. 2503-2555, doi: 10.1111/acfi.12981.
	Hunter, J.E. and Schmidt, F.L. (1990), <i>Methods of Meta-Analysis: Correcting Errors and Bias in Research Findings</i> , Sage, CA.
	Hunter, J.E. and Schmidt, F.L. (2014), <i>Methods of Meta-Analysis: correcting Error and Bias in Research Findings</i> , Sage Publications, Thousand Oaks, CA.
	Hunter, J.E., Schmidt, F.L. and Jackson, G.B. (1982), <i>Studying Organizations: Innovations in Methodology.</i> <i>Meta-Analysis: Cumulating Research Findings Across Studies</i> , Sage, Beverly Hills.
	IFAC (2020), "Creating value with integrated thinking: the role of professional accountants", available at: www.ifac.org/knowledge-gateway/audit-assurance/publications/creating-value-integrated-thinking (accessed 9 July 2024).
	International Integrated Reporting Council (IIRC) (2021), "International <ir> framework", available at: www.integratedreporting.org/wpcontent/uploads/2021/01/ InternationalIntegratedReportingFramework.pdf. (accessed 22 January 2023).</ir>
	Jensen, M.C. and Meckling, W.H. (1976), "Theory of the firm: managerial behavior, agency cost and capital structure", <i>Journal of Financial Economics</i> , Vol. 3 No. 4, pp. 305-360.
	Kannenberg, L. and Schreck, P. (2019), "Integrated reporting: boon or bane? A review of empirical research on its determinants and implications", <i>Journal of Business Economics</i> , Vol. 89 No. 5, pp. 515-567, doi: 10.1007/s11573-018-0922-8.
	Kepes, S., McDaniel, M.A., Brannick, M.T. and Banks, G.C. (2013), "Meta-analytic reviews in the organizational sciences: two meta-analytic schools on the way to MARS (the meta-analytic reporting standards)", <i>Journal of Business and Psychology</i> , Vol. 28 No. 2, pp. 123-143, doi: 10.1007/s10869-013-9300-2.
	Khelil, I. (2023), "Political connections and cost of debt: a meta-analysis", <i>Journal of Financial Reporting and Accounting</i> , doi: 10.1108/JFRA-11-2022-0413.
	Khlif, H., Ahmed, K. and Souissi, M. (2017), "Ownership structure and voluntary disclosure: a synthesis of empirical studies", <i>Australian Journal of Management</i> , Vol. 42 No. 3, pp. 376-403, doi: 10.1177/0312896216641475.
	Khlif, H., Hussainey, K. and Achek, I. (2015), "The effect of national culture on the association between profitability and corporate social and environmental disclosure: a meta-analysis", <i>Meditari Accountancy Research</i> , Vol. 23 No. 3, pp. 296-321, doi: 10.1108/MEDAR-12-2014-0064.

Khlif, H. and Souissi, M. (2010), "The determinants of corporate disclosure: a meta-analysis", <i>International Journal of Accounting and Information Management</i> , Vol. 18 No. 3, pp. 198-219, doi: 10.1108/18347641011068965.	Meditari Accountancy Research
*Kim, S., Maas, K. and Perego, P. (2017), "The effect of publication, format and content of integrated reports on analysts' earnings forecasts", available at: https://ssrn.com/abstract=2902549 (accessed 3 April 2023).	Kesedicii
Knapp, G. and Hartung, J. (2003), "Improved tests for a random effects meta-regression with a single covariate", <i>Statistics in Medicine</i> , Vol. 22 No. 17, pp. 2693-2710, doi: 10.1002/ sim.1482.	231
*Kosovic, K. and Patel, P. (2013), "Integrated reporting – Is it value relevant? A quantitative study on Johannesburg stock exchange", Jönköping International Business School, available at: www.diva-portal.org/smash/get/diva2:624126/FULLTEXT02 (accessed 3 April 2023).	
*Lee, K.W. and Yeo, G.H.H. (2016), "The association between integrated reporting and firm valuation", <i>Review of Quantitative Finance and Accounting</i> , Vol. 47 No. 4, pp. 1221-1250, doi: 10.1007/s11156-015-0536-y.	
*Leukhardt, L., Charifzadeh, M. and Diefenbach, F. (2022), "Does integrated reporting quality matter to capital markets? Empirical evidence from voluntary adopters", <i>Corporate Social Responsibility and Environmental Management</i> , Vol. 29 No. 5, pp. 1482-1494, doi: 10.1002/csr.2285.	
Leuz, C. and Verrecchia, R.E. (2000), "The economic consequences of increased disclosure", <i>Journal of Accounting Research</i> , Vol. 38, pp. 91-124, doi: 10.2307/2672910.	
Lipsey, M.W. and Wilson, D.B. (2001), Practical Meta-Analysis, Sage, CA.	
Liu, X., Vredenburg, H. and Steel, P. (2014), "A meta-analysis of factors leading to management control in international joint ventures", <i>Journal of International Management</i> , Vol. 20 No. 2, pp. 219-236, doi: 10.1016/j.intman.2013.07.001.	
López-López, J.A., Page, M.J., Lipsey, M.W. and Higgins, J.P.T. (2018), "Dealing with effect size multiplicity in systematic reviews and meta-analyses", <i>Research Synthesis Methods</i> , Vol. 9 No. 3, pp. 336-351, doi: 10.1002/jrsm.1310.	
*Loprevite, S., Ricca, B. and Rupo, D. (2018), "Performance sustainability and integrated reporting: empirical evidence from mandatory and voluntary adoption contexts", <i>Sustainability</i> , Vol. 10 No. 5, p. 1351, doi: 10.3390/su10051351.	
Lu, W. and Taylor, M.E. (2016), "Which factors moderate the relationship between sustainability performance and financial performance? A meta-analysis study", <i>Journal of International Accounting Research</i> , Vol. 15 No. 1, pp. 1-15, doi: 10.2308/jiar-51103.	
*Maama, H. and Marimuthu, F. (2022), "Integrated reporting and cost of capital in Sub-Saharan African countries", <i>Journal of Applied Accounting Research</i> , Vol. 23 No. 2, pp. 381-401, doi: 10.1108/JAAR-10-2020-0214.	
Majumder, M.T.H., Akter, A. and Li, X. (2017), "Corporate governance and corporate social disclosures: a meta-analytical review", <i>International Journal of Accounting and Information</i> <i>Management</i> , Vol. 25 No. 4, pp. 434-458, doi: 10.1108/IJAIM-01-2017-0005.	
*Martinez, C. (2016a), "Does the level of alignment with the integrated reporting framework reduce information asymmetry?", doi: 10.2139/ssrn.2684784 (accessed 3 April 2023).	
*Martinez, C. (2016b), "Effects of integrated reporting on the firm's value: evidence from voluntary adopters of the IIRC's framework", doi: 10.2139/ssrn.2876145 (accessed 3 April 2023).	
Maroun, W. (2019), "Does external assurance contribute to higher quality integrated reports?", <i>Journal of Accounting and Public Policy</i> , Vol. 38 No. 4, p. 106670, doi: 10.1016/j. jaccpubpol.2019.06.002.	
Maroun, W. (2022), "Corporate governance and the use of external assurance for integrated reports", <i>Corporate Governance: An International Review</i> , Vol. 30 No. 5, pp. 584-607, doi: 10.1111/corg.12430.	

MEDAR 32,7	*Matemane, R. and Wentzel, R. (2019), "Integrated reporting and financial performance of South African listed banks", <i>Banks and Bank Systems</i> , Vol. 14 No. 2, pp. 128-139, doi: 10.21511/bbs.14 (2).2019.11.
	Miller, C.C. and Cardinal, L.B. (1994), "Strategic planning and firm performance: a synthesis of more than two decades of research", <i>Academy of Management Journal</i> , Vol. 37 No. 6, pp. 1649-1665, doi: 10.5465/256804.
232	Minutiello, V. and Tettamanzi, P. (2022), "The quality of nonfinancial voluntary disclosure: a systematic literature network analysis on sustainability reporting and integrated reporting", <i>Corporate Social Responsibility and Environmental Management</i> , Vol. 29 No. 1, pp. 1-18, doi: 10.1002/csr.2195.
	Mokhtar, E.S. (2017), "Internet financial reporting determinants: a meta-analytic review", <i>Journal of Financial Reporting and Accounting</i> , Vol. 15 No. 1, pp. 116-154, doi: 10.1108/JFRA-07-2016-0061.
	Moloi, T. and Iredele, O. (2020), "Firm value and integrated reporting quality of South African listed firms", <i>Academy of Strategic Management Journal</i> , Vol. 19 No. 1, pp. 1-20.
	Ng, H.Y., Tronnes, P.C. and Wong, L. (2018), "Audit seasonality and pricing of audit services: theory and evidence from a meta-analysis", <i>Journal of Accounting Literature</i> , Vol. 40 No. 1, pp. 16-28, doi: 10.1016/j.acclit.2017.11.003.
	Nwachukwu, C. (2022), "Systematic review of integrated reporting: recent trend and future research agenda", <i>Journal of Financial Reporting and Accounting</i> , Vol. 20 Nos 3/4, pp. 580-598, doi: 10.1108/JFRA-10-2020-0308.
	*Obeng, V.A., Ahmed, K. and Cahan, S.F. (2021), "Integrated reporting and agency costs: international evidence from voluntary adopters", <i>European Accounting Review</i> , Vol. 30 No. 4, pp. 645-674, doi: 10.1080/09638180.2020.1805342.
	*Omran, M.S.Y., Zaid, M.A.A. and Dwekat, A. (2021), "The relationship between integrated reporting and corporate environmental performance: a green trial", <i>Corporate Social Responsibility and</i> <i>Environmental Management</i> , Vol. 28 No. 1, pp. 427-445, doi: 10.1002/csr.2059.
	Opare, S., Houqe, M.N. and Van Zijl, T. (2021), "Meta-analysis of the impact of adoption of IFRS on financial reporting comparability, market liquidity, and cost of capital", <i>Abacus</i> , Vol. 57 No. 3, pp. 502-556, doi: 10.1111/abac.12237.
	*Pavlopoulos, A., Magnis, C. and Iatridis, G.E. (2017), "Integrated reporting: is it the last piece of the accounting disclosure puzzle?", <i>Journal of Multinational Financial Management</i> , Vol. 41, pp. 23-46, doi: 10.1016/j.mulfin.2017.05.001.
	*Pavlopoulos, A., Magnis, C. and Iatridis, G.E. (2019), "Integrated reporting: an accounting disclosure tool for high quality financial reporting", <i>Research in International Business and Finance</i> , Vol. 49, pp. 13-40, doi: 10.1016/j.ribaf.2019.02.007.
	Permatasari, I. and Tjahjadi, B. (2023), "A closer look at integrated reporting quality: a systematic review and agenda of future research", <i>Meditari Accountancy Research</i> , Vol. 32 No. 3, doi: 10.1108/MEDAR-08-2022-1782.
	Pistoni, A., Songini, L. and Bavagnoli, F. (2018), "Integrated reporting quality: an empirical analysis", Corporate Social Responsibility and Environmental Management, Vol. 25 No. 4, pp. 489-507, doi: 10.1002/csr.1474.
	Pomeroy, B. and Thornton, D.B. (2008), "Meta-analysis and the accounting literature: the case of audit committee independence and financial reporting quality", <i>European Accounting Review</i> , Vol. 17 No. 2, pp. 305-330, doi: 10.1080/09638180701819832.
	*Raimo, N., Caragnano, A., Mariani, M. and Vitolla, F. (2022), "Integrated reporting quality and cost of debt financing", <i>Journal of Applied Accounting Research</i> , Vol. 23 No. 1, pp. 122-138, doi: 10.1108/JAAR-04-2021-0097.
	Ringquist, E.J. (2013), <i>Meta-Analysis for Public Management and Policy</i> , John Wiley and Sons, San Francisco.

Rizzato, F., Tonelli, A., Fiandrino, S. and Devalle, A. (2024), "Analysing SDG disclosure and its impact on integrated thinking and reporting", <i>Meditari Accountancy Research</i> , Vol. 32 No. 3, pp. 803-831, doi: 10.1108/MEDAR-07-2022-1751.	Meditari Accountancy
Rosenthal, R. (1979), "The file drawer problem and tolerance for null results", <i>Psychological Bulletin</i> , Vol. 86 No. 3, pp. 638-641, doi: 10.1037/0033-2909.86.3.638.	Research
Rosenthal, R. (1991), <i>Meta-Analytic Procedures for Social Research</i> , Applied Social Research Methods Series, 6 Book, Sage, Newbury Park.	233
Salehi, M., Fakhri Mahmoudi, M.R. and Daemi Gah, A. (2019), "A meta-analysis approach for determinants of effective factors on audit quality: evidence from emerging market", <i>Journal of</i> <i>Accounting in Emerging Economies</i> , Vol. 9 No. 2, pp. 287-312, doi: 10.1108/JAEE-03-2018- 0025.	
Salvi, A., Vitolla, F., Giakoumelou, A., Raimo, N. and Rubino, M. (2020), "Intellectual capital disclosure in integrated reports: the effect on firm value", <i>Technological Forecasting and Social</i> <i>Change</i> , Vol. 160, p. 120228, doi: 10.1016/j.techfore.2020.120228.	
Schreder, M. (2018), "Idiosyncratic information and the cost of equity capital: a meta-analytic review of the literature", <i>Journal of Accounting Literature</i> , Vol. 41 No. 1, pp. 142-172, doi: 10.1016/j. acclit.2018.07.001.	
*Senani, K.G.P., Ajward, R. and Kumari, J.S. (2022), "Determinants and consequences of integrated reporting disclosures of non-financial listed firms in an emerging economy", <i>Journal of Financial Reporting and Accounting</i> , doi: 10.1108/JFRA-03-2022-0083.	
*Serafeim, G. (2015), "Integrated reporting and investor clientele", Journal of Applied Corporate Finance, Vol. 27 No. 2, pp. 34-51, doi: 10.1111/jacf.12116.	
Setia, N., Abhayawansa, S., Joshi, M. and Wasantha Pathiranage, N. (2024), "Shifting perspectives: unveiling the dual nature of sustainability materiality in integrated reports", <i>Meditari Accountancy Research</i> , Vol. 32 No. 4, pp. 1291-1323, doi: 10.1108/MEDAR-08- 2023-2128.	
Siddiqui, S.S. (2015), "The association between corporate governance and firm performance–a meta- analysis", <i>International Journal of Accounting and Information Management</i> , Vol. 23 No. 3, pp. 218-237, doi: 10.1108/IJAIM-04-2014-0023.	
Song, F. (1999), "Exploring heterogeneity in meta-analysis: is the l'Abbe plot useful?", <i>Journal of Clinical Epidemiology</i> , Vol. 52 No. 8, pp. 725-730, doi: 10.1016/S0895-4356(99) 00066-9.	
Songini, L., Pistoni, A., Tettamanzi, P., Fratini, F. and Minutiello, V. (2022), "Integrated reporting quality and BoD characteristics: an empirical analysis", <i>Journal of Management and Governance</i> , Vol. 26 No. 2, pp. 579-620, doi: 10.1007/s10997-021-09568-8.	
Songini, L., Pistoni, A., Comerio, N. and Tettamanzi, P. (2023), "A decade of integrated reporting studies: state of the art and future research implications", <i>Accounting, Auditing</i> and Accountability Journal, Vol. 36 No. 9, pp. 226-252, doi: 10.1108/AAAJ-10-2021- 5490.	
*Soriya, S. and Rastogi, P. (2022), "The impact of integrated reporting on financial performance in India: a panel data analysis", <i>Journal of Applied Accounting Research</i> , Vol. 24 No. 1, doi: 10.1108/JAAR-10-2021-0271.	
*Soumillion, V. (2018), "The value relevance of integrated reporting in South Africa", available at: https://libstore.ugent.be/fulltxt/RUG01/002/509/073/RUG01-002509073_2018_0001_AC.pdf (accessed 17 August 2023).	
Spence, M. (1973), "Job market signaling", <i>The Quarterly Journal of Economics</i> , Vol. 87 No. 3, pp. 355-374, doi: 10.2307/1882010.	
Spence, M. (2002), "Signaling in retrospect and the informational structure of markets", <i>American Economic Review</i> , Vol. 92 No. 3, pp. 434-459, doi: 10.1257/00028280260136200.	

MEDAR 32,7	*Sriani, D. and Agustia, D. (2020), "Does voluntary integrated reporting reduce information asymmetry? Evidence from Europe and asia", <i>Heliyon</i> , Vol. 6 No. 12, p. e05602, doi: 10.1016/j. heliyon.2020.e05602.
	Stanley, T.D. (2005), "Beyond publication bias", <i>Journal of Economic Surveys</i> , Vol. 19 No. 3, pp. 309-345, doi: 10.1111/j.0950-0804.2005.00250.x.
234	Stanley, T.D., Doucouliagos, H., Giles, M., Heckemeyer, J.H., Johnston, R.J., Laroche, P., Nelson, J.P., Paldam, M., Poot, J., Pugh, G., Rosenberg, R.S. and Rost, K. (2013), "Meta-analysis of economics research reporting guidelines", <i>Journal of Economic Surveys</i> , Vol. 27 No. 2, pp. 390-394, doi: 10.1111/joes.12008.
	Stefanescu, C.A. (2022), "Linking sustainability reporting frameworks and sustainable development goals", <i>Accounting Research Journal</i> , Vol. 35 No. 4, pp. 508-525, doi: 10.1108/ARJ-07-2020-0196.
	Stubbs, W. and Higgins, C. (2014), "Integrated reporting and internal mechanisms of change", Accounting, Auditing and Accountability Journal, Vol. 27 No. 7, pp. 1068-1089, doi: 10.1108/ AAAJ-03-2013-1279.
	*Sun, Y., Qiao, X., An, Y., Fang, Q. and Wu, N. (2022), "Does multiple capitals disclosure affect the capital market? An empirical analysis in an integrated reporting perspective", <i>Frontiers in Psychology</i> , Vol. 13, p. 837209, doi: 10.3389/fpsyg.2022.837209.
	Velte, P. (2019), "What do we know about meta-analyses in accounting, auditing, and corporate governance?", Meditari Accountancy Research, Vol. 27 No. 1, pp. 17-43, doi: 10.1108/MEDAR- 03-2018-0317.
	Velte, P. (2022), "Archival research on integrated reporting: a systematic review of main drivers and the impact of integrated reporting on firm value", <i>Journal of Management and Governance</i> , Vol. 26 No. 3, pp. 997-1061, doi: 10.1007/s10997-021-09582-w.
	Velte, P. and Stawinoga, M. (2017), "Integrated reporting: the current state of empirical research, limitations and future research implications", <i>Journal of Management Control</i> , Vol. 28 No. 3, pp. 275-320, doi: 10.1007/s00187-016-0235-4.
	Veltri, S. and Silvestri, A. (2020), "The value relevance of corporate financial and nonfinancial information provided by the integrated report: a systematic review", <i>Business Strategy and the Environment</i> , Vol. 29 No. 8, pp. 3038-3054, doi: 10.1002/bse.2556.
	Verrecchia, R.E. (1983), "Discretionary disclosure", <i>Journal of Accounting and Economics</i> , Vol. 5, pp. 179-194, doi: 10.1016/0165-4101(83)90011-3.
	Vitolla, F., Raimo, N. and Rubino, M. (2019), "Appreciations, criticisms, determinants, and effects of integrated reporting: a systematic literature review", <i>Corporate Social Responsibility and Environmental Management</i> , Vol. 26 No. 2, pp. 518-528, doi: 10.1002/csr.1734.
	*Vitolla, F., Salvi, A., Raimo, N., Petruzzella, F. and Rubino, M. (2020), "The impact on the cost of equity capital in the effects of integrated reporting quality", <i>Business Strategy and the Environment</i> , Vol. 29 No. 2, pp. 519-529, doi: 10.1002/bse.2384.
	Wahl, A., Charifzadeh, M. and Diefenbach, F. (2020), "Voluntary adopters of integrated reporting– evidence on forecast accuracy and firm value", <i>Business Strategy and the Environment</i> , Vol. 29 No. 6, pp. 2542-2556, doi: 10.1002/bse.2519.
	Walker, E., Hernandez, A.V. and Kattan, M.W. (2008), "Meta-analysis: its strengths and limitations", <i>Cleveland Clinic Journal of Medicine</i> , Vol. 75 No. 6, pp. 431-439.
	Wang, K.T. and Shailer, G. (2018), "Does ownership identity matter? A meta-analysis of research on firm financial performance in relation to government versus private ownership", <i>Abacus</i> , Vol. 54 No. 1, pp. 1-35, doi: 10.1111/abac.12103.
	*Wen, L.P., Yap Kiew Heong, A. and Lim Chee Hooi, S. (2017), "Integrated reporting and financial performance: evidence from Malaysia", <i>Management and Accounting Review (MAR)</i> , Vol. 16 No. 2, pp. 101-130.

*Wu, Y. and Zhou, S. (2022), "Do firms practicing integrated reporting engage in less myopic behavior? International evidence on opportunistic earnings management", <i>Corporate Governance: An</i> <i>International Review</i> , Vol. 30 No. 3, pp. 290-310, doi: 10.1111/corg.12401.	Meditari Accountancy
*Zhou, S., Simnett, R. and Green, W. (2017), "Does integrated reporting matter to the capital market?", <i>Abacus</i> , Vol. 53 No. 1, pp. 94-132, doi: 10.1111/abac.12104.	Research
Zhou, S., Simnett, R. and Hoang, H. (2019), "Evaluating combined assurance as a new credibility enhancement technique", AUDITING: A Journal of Practice and Theory, Vol. 38 No. 2, pp. 235-259, doi: 10.2308/ajpt-52175.	235
*Zúñiga, F., Pincheira, R., Walker, J. and Turner, M. (2020), "The effect of integrated reporting quality on market liquidity and analyst forecast error", <i>Accounting Research Journal</i> , Vol. 33 Nos 4/5, pp. 635-650, doi: 10.1108/ARJ-07-2019-0145.	

Corresponding author

Filippo Zanin can be contacted at: filippo.zanin@uniud.it